

Strathmore University SU + @ Strathmore University Library

Electronic Theses and Dissertations

2019

Effectiveness of the maintenance practices of service providers on the functionality of rural water supply facilities in Turkana County

James O. Otieno Strathmore Business School (SBS) Strathmore University

Follow this and additional works at https://su-plus.strathmore.edu/handle/11071/9973

Recommended Citation

Otieno, J. O. (2019). Effectiveness of the maintenance practices of service providers on the

functionality of rural water supply facilities in Turkana County [Thesis, Strathmore

University]. https://su-plus.strathmore.edu/handle/11071/9973

This Thesis - Open Access is brought to you for free and open access by DSpace @Strathmore University. It has been accepted for inclusion in Electronic Theses and Dissertations by an authorized administrator of DSpace @Strathmore University. For more information, please contact <u>librarian@strathmore.edu</u>

Effectiveness of the Maintenance Practices of Service Providers on the Functionality of Rural Water Supply Facilities in Turkana county

James Origa OTIENO

MPPM/96532/2016

Submitted in partial fulfilment of the requirements for the Degree of Master of Public Policy and Management at Strathmore University

Strathmore University Business School

STRATHMORE UNIVERSITY, Nairobi, Kenya

JUNE 2019

This dissertation is available for Library use on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

i

DECLARATION

I declare that this work has not been previously submitted and approved for the award of a degree by this or any other University. To the best of my knowledge and belief, the thesis contains no material previously published or written by another person except where due reference is made in the thesis itself.

©No part of this dissertation may be reproduced without the permission of the author and Strathmore University

James Origa Otieno June 2019

Approval

The dissertation of James Origa Otieno was reviewed and approved by:

Dr. Monica Kerretts-Makau (Supervisor) Strathmore Business School Strathmore University

Dr. George N. Njenga Executive Dean, Strathmore University Business School

Professor Ruth Kiraka Dean, School of Research & Graduate Studies

ABSTRACT

The purpose of this study was to assess the effects of post-construction infrastructure maintenance practices of rural water service providers in Turkana county on the functionality of the facilities. Data on prevalent maintenance practises and attendant revenue mobilisation and management practises was collected from ten sampled sites within Turkana county through quantitative household questionnaires. Focused group discussions with the service providers and key informant interviews with county government officials and other water sector partners was also carried out to validate the quantitative data. The data was analysed using descriptive statistics and instrumental variables regression analysis. The study findings showed endogeneity in the independent variables with the time taken to repair broken down water facilities emerging as the main predictor of functionality. The regression results indicate that a rural water system in Turkana is likely to be functional if it takes less time to repair compared to the base category of less than 24 hours. Affordability of water by the households emerged as another key factor that determines functionality since it influences sufficiency of revenues needed to respond to a break down. The results demonstrate a cyclic relationship in which affordability increases functionality as more households are able to and willing to contribute towards repair costs, hence making funds available for repairs and increasing prompt response to break downs. In turn, functionality increases household's willingness to pay since the system is well maintained thus available when needed.

STRATHMORE UNIVERSITY LIBRARY SPECIAL COLLECTIONS

TABLE OF CONTENTS

DECLARATION
ABSTRACTiii
TABLE OF CONTENTSiv
LIST OF FIGURES
LIST OF TABLES
LIST OF ABBREVIATIONSix
ACKNOWLEDGEMENT
SPECIAL DEDICATION
Chapter 1 : INTRODUCTION
1.1 Introduction
1.2 Background to the study
1.3 Problem definition
1.4 Research Objectives
1.5 Research Questions
1.6 Scope and limitations of the study
1.7 Significance of the study
Chapter 2 : LITERATURE REVIEW
2.1 Introduction
2.2 Theoretical Review
2.2.1 Life-Cycle Service-delivery theory
2.3 Empirical review

2.3.1 Sustainability and functionality of rural water supply schemes	8
2.3.2 Functionality as an indicator of sustainability	10
2.3.3 Maintenance and repair practices in rural water supply delivery	
2.3.4 Financing and revenue mobilization for maintenance	12
2.4 Literature review summary and research gaps	14
2.5 Conceptual framework and definition of study variables	15
Chapter 3 : RESEARCH METHODOLOGY	17
3.1 The research design and rationale	17
3.2 Sampling	17
3.2.1 Sampling frame and procedure	17
3.2.2 Sample size	18
3.3 Data collection tools	21
3.4 Data analysis	
3.5 Research Quality - validity, reliability and objectivity of the research	21
 3.5 Research Quality – validity, reliability and objectivity of the research. 3.6 Ethical considerations during the study	21
 3.5 Research Quality – validity, reliability and objectivity of the research. 3.6 Ethical considerations during the study Chapter 4 : PRESENTATION OF RESEARCH FINDINGS 	21
 3.5 Research Quality – validity, reliability and objectivity of the research. 3.6 Ethical considerations during the study . Chapter 4 : PRESENTATION OF RESEARCH FINDINGS	21 21 22 22
 3.5 Research Quality – validity, reliability and objectivity of the research. 3.6 Ethical considerations during the study . Chapter 4 : PRESENTATION OF RESEARCH FINDINGS	21 21 22 22 22 22
 3.5 Research Quality – validity, reliability and objectivity of the research. 3.6 Ethical considerations during the study Chapter 4 : PRESENTATION OF RESEARCH FINDINGS	21 21 22 22 22 22
 3.5 Research Quality – validity, reliability and objectivity of the research. 3.6 Ethical considerations during the study. Chapter 4 : PRESENTATION OF RESEARCH FINDINGS	21 21 22 22 22 22
 3.5 Research Quality – validity, reliability and objectivity of the research. 3.6 Ethical considerations during the study	21 21 22 22 22 22

5.1 Introduction	
5.2 Summary discussions of the study findings	
5.3 Analysis of the study findings within the context of other literature	41
Chapter 6 : CONCLUSIONS AND RECOMMENDATIONS	44
6.1 Conclusions	44
6.2 Policy significance of the study findings	45
6.3 Recommendations	47
6.4 Suggestions for further research	
REFERENCES	50
APPENDICES	i
APPENDIX A: ETHICAL CLEARANCE FOR DATA COLLECTION	i
APPENDIX B: DATA COLLECTION TOOLS	ii

LIST OF FIGURES

Figure 2-1 Study conceptual framework
Figure 4-1 Gender distribution of the study respondents
Figure 4-2: Causes of water facilities breakdown in Turkana
Figure 4-3: Time taken to respond to water facility breakdowns in Turkana
Figure 4-4: Perceptions on who should carry out water facilities repair in Turkana 27
Figure 4-5: Ease of access to spare parts in Turkana
Figure 4-6: How service providers in Turkana obtain spare parts
Figure 4-7: Sources and kinds of post-construction support in Turkana
Figure 4-8: Perceptions on the role of county government of Turkana in rural water supply
delivery
Figure 4-9: Challenges in technical support and monitoring of rural water service delivery
in Turkana
Figure 4-10: Sources of revenue for O & M in Turkana
Figure 4-11: Factors influencing water users' willingness to pay for water in Turkana 34
Figure 4-12: How revenues are used by rural water service providers in Turkana
Figure 4-13: Regression analysis results
Figure 4-14: Results of testing the fitness of the regression model
Figure 4.15: Repute of checking for completion among mudictor conichies 27

LIST OF TABLES

Table 3-1 Summary of the sampling procedure	9
Table 3-2:The 10 Initially selected primary sampling units and sizes	9
Table 3-3: The 10 finally selected primary sampling units during field work 2	20
Table 4-1 Typologies and frequencies of maintenance activities by service providers i	in
Turkana	24
Table 4-2: Average cost of water in various communities in Turkana	3
Table 4-3: Summary of regression results 3	38

LIST OF ABBREVIATIONS

.

WHO	World Health Organisation
UN	United Nations
UNICEF	United Nations International Children's Fund
GDP	Gross Domestic Product
MDG	Millennium Development Goals
SDG	Sustainable Development Goals
WMC	Water Management Committee
WSP	Water Service Provider
WASREB	Water Services Regulatory Board
NGO	Non-Governmental Organisation
0 & M	Operations and Maintenance

ACKNOWLEDGEMENT

I would like to express my sincere gratitude to my supervisor Dr. Monica Kerretts for her advice and guidance throughout developing the thesis proposal to writing the final thesis. I would also like to thank my lecturers at the Strathmore University Business School who laboured to provide me with theoretical backgrounds, conceptual frameworks and practical analytical skills in public policy and public leadership affairs.

I am most grateful to my wife, Eve Origa, for her constant encouragements and understanding throughout my studies at Strathmore university.

SPECIAL DEDICATION

With deep sorrow in my heart, I dedicate this thesis to my MPPM 2016 cohort classmate, the late Isaack Mutuma Kaaria. A great friend to us all and loving husband to Loise. You left us too soon, just before graduation. It would have been so lovely for us to graduate together from Strathmore University and continue doing what you loved most-serving the nation faithfully and with integrity. I will miss having the conversations with you on how we can better the public leadership in our beloved country, having been imparted with the skills and strategy of public policy making. The nation has lost a great servant leader. Till we meet again brother, rest in peace.

Chapter 1 : INTRODUCTION

1.1 Introduction

Despite the United Nations (UN) member countries declaring access to safe water and sanitation a basic human right (UN, 2010), recent reports by the World Health Organization (WHO) & The United Nations Children's Fund (UNICEF), 2017) indicate that 159 million people,58% of them living in sub-Saharan Africa, still rely on unsafe water sources putting them at high risk of premature mortality and preventable morbidity. The Global Risks report also indicates the global water crisis where millions of people are without access to improved water sources as the number one threat facing the planet in the coming decades (World Economic Forum, 2015). To achieve water security especially for the millions unserved in sub-Saharan Africa, an increased investment in new water infrastructure to expand coverage as well as institutional strengthening for sustainable services delivery are key. This emphasizes the critical importance of the country being able to sustainably manage existing water resources as well as ensuring sustainable and reliable water services delivery to contain water looses in an already resource constraint context.

Water supply projects are designed to achieve three outcomes. First is an ultimate outcome of continuous functionality of installed facilities. Second is sustainable management of infrastructure services including sustainable revenue streams to recover costs of operations and maintenance, maximised efficiency and professionalism in service delivery and management of risks facing source over-abstraction. Lastly is affordability and equitable access particularly for those having less ability to pay within the community. The practises adopted by the service providers mandated to operate and maintain the installed systems is thus critical for successful attainment of these three outcomes. This study sought to assess the effects of infrastructure maintenance practises adopted by the rural water service providers responsible for operating the rural water schemes in Turkana county on their long-term functionality as the ultimate outcome.

This thesis is organised into 6 chapters. Chapter 1 presents a contextual background of the proposed study detailing the rationale, objectives, limitations and significance of the study. In chapter 2, a review of literature on sustainability of rural water services and

management of water services delivery, particularly on maintenance of water infrastructure and financing the maintenance needs. A conceptual framework is developed at the end of chapter 2 to guide the study research design and development of data collection instruments. Chapter 3 details the research design and methods applied including the approach chosen for sample selection, data collection. Chapter 4 presents the findings of the study from the field data. Chapter 5 presents a discussion of the study findings as related to the literature on maintenance practices in water services delivery, revenue management practices as well as rural water sustainability and functionality as discussed in chapter two. Chapter 6 presents the study conclusions, significance of this study's findings to policy makers as well as highlights the limitation of the study and make suggestions for further research.

1.2 Background to the study

In response to public infrastructure funding gaps, increasing stresses on water sources exacerbated by climate change effects, distributional inequalities in access to safe water services and the worrying high infrastructure failure rates (Castro & Heller, 2012), three trends characterise the current global water sector. First is increasing focus on strengthening and professionalizing service delivery institutions (Fogelberg, 2013; Lockwood & Smits, 2011; Rural Water Supply Network, 2010; Moriarty, Smits, Butterworth, & Franceys, 2013; Schouten & Moriarty, 2013; Hepworth, 2016; Hope, Foster, Money, & Rouse, 2012; Laban, 2007); second is a move towards adaptive ecosystem management to manage competing demands for multiple water uses (Boelee, Chiramba, & Khaka, 2011; Honkonen, 2017; Martínez, 2014; Varady, Zuniga-Teran, Garfin, Martín, & Vicuña, 2016) and thirdly is increased efforts to leverage the constrained public funds with private sector financing and expertise. (Castro, 2018; Fonseca & Pories, 2017; James, Trémolet, & Ikeda, 2016; Müller, 2016).

For more than three decades after gaining independence in 1963 until 1999, the government of Kenya operated without a substantive strategy and policy for the water sector. While the 2002 and 2016 water sector reforms in Kenya has led to great advances in improving professional management of urban water services, success in the

management of rural water services remains characterised by poor management practises and low sustainability of installed infrastructure. The Kenya water sector regulator, the Water Services Regulatory Board's (WASREB) interventions to develop regulatory tools to ensure sustainable management of rural water services (WASREB, 2018) are yet to yield the expected results for residents of remote rural contexts, especially the dry northern Kenya counties.

The rural water services delivery chain involves organising the production and supply of services to end users. These services include facility maintenance to keep the infrastructure in good working condition as well as attendant collection and management of revenues to cover the cost of facility maintenance. Ultimate functionality of the facilities is a sum-total of both services. The service providers responsible for the management of these facilities are expected to have sufficient capacity to maintain certain level of professional service delivery that adds public value to the water users. At the service level, the managing entity, whether it's a large water utility or a small communitybased operator, is expected to carry out daily regular maintenance such as simple cleaning of facilities; prepare and follow an effective preventive maintenance regime involving acquiring of spare parts, scheduled replacement of parts; promptly responding to facility breakdown within the shortest time possible; Administrative tasks such as keeping of records and monitoring key trends to enable accountability to regulatory authorities; financial management such as collection of rates, deployment of collected rates to efficiently meet the cost of operating and maintaining the system as well as having a responsive customer engagement interface and processing of emerging complaints.

WASREB requires every water service provider to organise a maintenance system that ensures the assets are in operating conditions that enables continuity of services. The subject of inquiry in this study was to examine to what extent service providers in rural settings with small point water sources such as Turkana are organising such effective maintenance regimes.

1.3 Problem definition

The Government of Kenya's goal is to achieve 100% water access by the year 2030 to all citizens. To this end, the government invests about Ksh.40 billion annually in developing new water supply infrastructure country-wide ranging from large capacity dams. pipelines and boreholes (WASREB, 2018). Turkana county has been a priority county for increased infrastructural investment owing to the low water access coverage of only 39% compared to a national average of 55% (Mwangi, 2013; Turkana county Government, 2016). A public expenditure review by the government of Kenya indicates that the Turkana county government has in the last 3 years, 2015/16 to 2017/18, cumulatively invested about Ksh.347 million in developing new water infrastructure (Government of Kenya, 2018). Despite these efforts to increase investments in new infrastructure, a study by Oxfam mapping existing water supply schemes in Turkana county observed that 33% of these water sources were non-functional and not delivering services as designed (Oxfam, 2017). This is happening despite the existence of water management committees responsible for collecting water use revenues and using these funds for maintenance operations to ensure these water points are operating at full potential. Similar findings on rural water supply scheme functionality status have been reported by Kwena & Moronge (2015), Lockwood & Smits (2011), Sutton (2004), Government of Kenya (2009) and the United Republic of Tanzania (2016) who observed that between 30-45% of rural water supply systems in Africa will be non-functional within 3-5 years after construction. This outcome raises concerns over the effectiveness of the management systems put in place to ensure sustainability of installed infrastructure. Unless facility functionality is addressed and significantly improved, the 100% universal access rate envisioned will most likely not be achieved.

Studies by Leclert, Nzioki, & Feuerstein (2016); Walters & Javernick-Will (2015); Skinner (2009) and Rural Water Supply Network (2010) all suggest that the critical underlying causes of this high non-functionality rates of rural water facilities are weakness in revenue management, ineffective maintenance activities and weak social accountability structures lacking in transparency, responsiveness and accountability of water service providers. Even the government of Kenya, in the draft 2018 national water policy statement, recognizes that operators of rural water supply systems operate with a weak governance framework, low technical capacity, insufficient revenues, and application of inappropriate technology (Ministry of Water,2018). These challenges, the ministry recognizes, affect the effectiveness of their service delivery and ultimate functionality of water facilities.

This study sought to assess the maintenance practises of the rural water management committees responsible for operating the water points in Turkana county and the effects these practises have on water schemes functionality. The findings will form an evidence basis for proposing appropriate policy interventions in operating rural water schemes such that high functionality is achieved.

1.4 Research Objectives

The **general objective** of this study was to assess the effects of infrastructure maintenance practises of water service providers responsible for operating the rural water schemes in Turkana county on their functionality.

The specific objectives were to:

- 1. To examine the maintenance practices of water service providers managing rural water schemes within Turkana county.
- To investigate how the service providers managing rural water schemes in Turkana county raise and manage revenues needed to facilitate the maintenance needs.
- To establish how the maintenance practices and attendant revenue management practises of the water service providers affect the functionality of the water schemes.

1.5 Research Questions

The study research questions were as follows;

1. What are the specific types of infrastructure maintenance activities practiced by

rural water service providers in Turkana county?

- 2. What are the specific ways rural water service providers in Turkana county raise and manage revenues needed to facilitate the maintenance of the water schemes?
- 3. How do the maintenance and revenue management practises of the rural water service providers in Turkana county affect the functionality of the water schemes?

1.6 Scope and limitations of the study

The study focused on maintenance practises of the water service providers and the attendant revenue management and their effectiveness in ensuring functionality of the schemes. The study is limited to management of water service delivery and its effects on long-term operational functionality and does not extend to effects of climate and governance of water services.

1.7 Significance of the study

The outcome of this study, targeted for dissemination to national and county government political leaders and senior level bureaucrats as well as non-governmental agencies supporting rural water services delivery, is envisioned to provide critical evidence needed to trigger changes in how rural water services provision are organised and financed. Policymakers will find the outcomes of this study useful in guiding policy proposals aimed at re-designing the institutional architecture for rural water services delivery as well as ensuring equitable access to water for all. The findings will also support Non-governmental agencies and water users/citizen action groups in refining the specific role they need to play in ensuring that rural water services are sustainably maintained.

Chapter 2 : LITERATURE REVIEW

2.1 Introduction

This chapter presents a review of literature on management of water services delivery, operations and maintenance practises, financing of maintenance needs as well as rural water sustainability and functionality. Based on the extensive literature review, a conceptual framework was then developed at the end of the literature review to guide the design of the data collection instruments applied in operationalizing the study variables and answering the research questions.

2.2 Theoretical Review

2.2.1 Life-Cycle Service-delivery theory

Lockwood & Smits (2011) defines rural water supply service delivery as "an approach to the provision of rural water supply services, which emphasises the entire life-cycle of a service, consisting of both the hardware and software required to sustain a certain level of service". This definition diverges from the traditional narrow infrastructure development approach, as it draws attention to the full life-cycle of an infrastructure service which emphasizes post-construction technical support, planning for longer-term facilities maintenance and expansions. Lockwood, Smit, Schouten, & Moriarty (2010) further expounded that a rural water supply service delivery model is: "the how to of applying the service delivery approach and describes the policy, legal, institutional, financial, governance and normative frameworks that determine what services will be provided to the consumers and how this will be done".

Service delivery models may include different management arrangements such as selfsupply referring to water supplies developed largely or wholly through user owninvestment usually at household level (Mekonta, Butterworth, & Holtslag, 2015); Community-based management model-a model in which the communities collectively manage, maintain and operate their own supply systems through selection of a voluntary water management committee by community water users (SNV, 2010); Private operator supply model where private sector actor is contracted to operate and manage the service delivery and is remunerated based on pre-determined performance indicators; Utility management model where a public utility is primarily responsible for the management of the services and lastly Municipal management model where a municipal water department is in charge of the operation and maintenance of the systems (World Bank Group, 2017; RWSN Executive Steering Committee, 2010). Lockwood et al., (2010) asserts that it is "difficult or indeed impossible to conceptualise one 'generic' model, which can be applied universally".

Lockwood (2014) presented that at the service provider level, the key activities to be carried out by the entities managing the water supply services involve day-to-day system operations such as regular maintenance, cleaning, provision of spare parts, regular replacement of consumable parts of the equipment; Administrative tasks involving keeping records; Financial management involving the practices of revenue collection, calculation of income & expenditures, maintenance of bank accounts as well as consumer interface involving management of complains and feedback from water users.

The examination of the maintenance practises and attendant revenue mobilisation and management practises in Turkana county was guided by the emerging theory advancing a service delivery and life-cycle thinking paradigm as described above. The study examined the extent to which the service providers in Turkana have adopted the defined maintenance and revenue management practises as specified in the service delivery approach theory.

2.3 Empirical review

2.3.1 Sustainability and functionality of rural water supply schemes

Hope, Pérez-Foguet, Katomero, & Georgiadou (2008) warns that the rush to meet the sustainable development goals (SDGs) of universal access by 2030 through increasing infrastructure coverage and accelerated development of new water supply schemes stands

the risk of diverting attention from a focus on beneficiaries' involvement, building institutional capacity and ensuring sufficient operations and maintenance systems which are crucial for sustaining services. Sustainability requires that to keep a water system functioning over the long-term, the dynamic and systemic interactions between technical, social, financial, institutional, and environmental factors that can lead to premature water system failure must be concurrently addressed as a precursor for infrastructure development (Walters & Javernick-Will, 2015).

The topic of sustainability of rural water supply services has been extensively studied empirically. Several authors have endeavoured to explain the multiple dimensions of sustainability such as social, environmental, financial and technical dimensions. Travis & Sara (1997) and Ostrom (2000) analysis concluded that what matters most for the social sustainability of rural water points involves working with and having the community members agree on infrastructure options, ascertain the community's preferences for service levels, and clarify the community's responsibilities and preferences for financing, operations and maintenance structures. Peter & Nkambule (2012); Enéas da Silva, et al., (2013); Jones, et al.,(2013), Jones, Anya, Stacey, & Weir, (2012), Amjad et al., (2015) conclusions from both literature review and empirical analysis emphasize technical sustainability-defined as the availability of equipment and technically skilled people for operating the system, financial sustainability defined as capacity to generate sufficient revenues as well as environmental sustainability defined as capacity of the water source to continuously provide water without drying up as critical factors affecting functionality of rural water points.

Chowns (2014), from extensive quantitative and qualitative analysis of the community management model of 679 rural water points in Malawi concluded that the model has benefited more donors "as a means of offloading responsibility for public service provision" but has failed in sustaining the benefits from installed capital infrastructure for communities. She describes two broad dimensions of sustainability. Environmental sustainability as the ability of the natural ecosystem and resources to provide services without compromising the future availability of those resources and economic sustainability as implying outcomes will continue after capital inputs cease, or that new

resources are found to replace the initial inputs. From her conclusions, the sustainability failure has been largely economic contributed to by the combined effects of weak accountability in financial management as well as low technical capacity of rural service operators to maintain the schemes in good condition. In her recommendations, she reinforces Lockwood and Smits (2011) that by taking a service delivery approach rather than a project approach to rural water supply, the focus essentially shifts away from delivering stand-alone outputs measured as completed water points to delivering outcomes measured as continued access to clean water. Montgomery, Bartram, & Elimelech (2009) on the other hand, concluded that the three most critical sustainability factors are effective community demand, sufficiency of local financing and cost recovery, and existence of a dynamic operation and maintenance regime. They observed that it is the interaction of these factors that ultimately affects how well an installed water scheme functions over time.

Walters' (2015) study used a detailed Delphi survey and cross impact questionnaire study to investigate the interaction of factors influencing functionality of rural water systems by modelling the factors as a system. His results concluded that the factors, including ability of the government to provide post-construction maintenance expertise and resources, existence of a water system management entity that collects and deploys water rates for maintenance, availability of spare parts and ability of the water source to sustain abstraction among others, interacted as a complex system. In his conclusions, he emphasized the lynchpin of sustainable management of rural water schemes as operationalising a system that '*aptly considers all these complexities*''.

2.3.2 Functionality as an indicator of sustainability

Lockwood & Le Gouais (2014) and the World Bank (2017) in using functionality as a proxy indicator of sustainability, both argue against the definition of functionality as a one-off binary check on a water facility to determine whether the system is working or not working at a single point in time. Similar sentiments are expressed by Leclert (2012); Tincani et al. (2015); Carter and Ross (2016) who argue that functionality should be tracked over time to give a picture of sustainability of the water facility. Bonsor,

MacDonald, Casey, Carter, & Wilson (2018) decry the lack of a single accepted definition of constitutes a functioning water point.

In an experimental research in Ghana, they introduced a tiered approach for defining functionality of water points fitted with hand-pumps. They thus defined a functional water facility as one that is physically working and producing water at the time of the survey visit, provides the minimum design yield and has less than 30 days of downtime within the last 12 months. Adank et al., (2013) used a stroke and leakage test to determine the functionality of water points. They set that handpump water points are functional if they pass both tests while those that pass only one of the two tests are classified as partially functional. Those that pass neither tests are classified as broken-down systems. The stroke test involved "taking a maximum of 40 strokes, administered within one minute, to fill the bucket for Afridev and India Mark II and 30 strokes for Nira AF-85 hand pumps" while the leakage test involved "resuming pumping after five minutes rest following the stroke test. If water flows from the hand pump within five strokes, the pump has passed the leakage test". While this definition seems promising and practical, its disadvantage is that its only limited to handpump water sources. It leaves out many more water sources common in the rural areas such a diesel generator pumping sets. Langdown (2018), in a qualitative evaluation of water points in northern Kenya, provides a more transient definition of functionality based on the downtime and number of days the water point has been operating in the preceding fourteen days. He concluded that a water point should be considered functional if it didn't break down for more than 24 hours not more than three times during the preceding two-week period, otherwise it should be considered nonfunctional.

In this study, Langdown's definition of functionality was applied given its strength in providing a reliability-based transient picture rather than a static picture of functionality as well as its applicability to any type of water source.

2.3.3 Maintenance and repair practices in rural water supply delivery

Harvey & Reed (2007) asserted that irrespective of the management model applied-be it community management or utility management, the key tasks of the service provider is

essentially to set and collect water tariffs and use these to maintain and repair the water supply facilities and where enabling expand the facilities. The Ghana community water boards managing rural water points used financial and operational indicators to assess the sustainability of rural water points (International Water and Sanitation Centre (IRC), 2012). From a financial indicators position, they defined a sustainable water point as one whose annual income exceeds annual expenditure, the service providers maintain proper revenue records, operate a bank account, sets water tariffs based on projected costs of operations and maintenance and carries out an audit at least once a year. Based on operational indicators, a sustainable water point was defined as one that has spare parts and mechanics in less than 24 hours, corrective maintenance practises in response to a facility breakdown is completed in less than 24 hours and at least one periodic maintenance or system overhaul is completed in a year.

Simukonda, Farmani, & Butler (2018) in emphasizing the importance of proper maintenance, assert that the performance of water supply infrastructure depends on the maintenance regime adopted. They concluded that the main causes of having a weak maintenance regime are insufficient funds due to poor revenue collection, poor data management and a lack of sufficient technical and managerial skills among service providers. Behailu, Hukka, & Katko (2016) in a detailed analysis of the causes of failures of rural water schemes in Ethiopia, concluded that lack of timely maintenance, failure of rehabilitation, lack of spare part supplies, and inadequate cost recovery are the key causes apart from environmental causes such as droughts. Boulenouar & Schweitzer (2015) study asserted that well planned, resourced and effectively implemented maintenance regimes for rural water points helps avoid catastrophic lump sum expenditures needed to replace failed components and associated premature failure and frequent breakdowns.

2.3.4 Financing and revenue mobilization for maintenance

While Bohm, Essenburg, & Fox (1993) concluded that rural water supply systems are not sustainable unless grants are available to finance most of the initial construction costs, Kaliba, Norman, & Chang (2003) asserts that water systems will only be sustainable in

the long-term if water users are willing to pay user charges sufficient to cover the costs of delivering the services. This section considers past research and theoretical literature on revenue generation required for water infrastructure capital development, maintenance and factors related to consumers' payments for water services.

Central to the successful delivery of rural water services is revenue mobilization and management specially to cover for the continuous operations and maintenance of the facilities. Chowns (2014) demonstrated, from a study of 679 water points in Malawi, that the ability and willingness to pay for recurrent operations and maintenance costs in rural water supply is the most significant factor in ensuring sustainability. Adank et al., (2013) studied the status of water supply in three rural districts in Ghana and found out that inability to raise funds for continuous Operation and maintenance was considered the major reason for unreliability in services. This lack of funds, they observed, impeded the ability of the service providers to maintain water facilities as money realized is inadequate to buy spare parts, properly train and provide competitive salaries to attract high caliber technical personnel. Carter, Harvey, & Casey (2010) demonstrated that in most cases, the revenues collected from water use fees were often much lower and inadequate to meet these recurrent costs. As such, communities are often left to solicit for financial support from external agencies including NGOs or local government offices (Davis et al. 2008). Several factors have been put forth as affecting water users' willingness and ability to make payments and thus sustainable resource mobilization practices of water service provider's managing those water points. Key among these is service delivery levels.

A study by Koehler, Thomson, & Hope (2015) observed that payments made by water users in rural areas is contingent on level of services provided. They observed that the higher the quality of services and reliability of water supply, the higher the revenues mobilized from the water users. Similarly, Shah & Mills (2018) highlighted the vicious cycle relating rural water facilities failure and the water users' willingness to pay. They posited that even the poorest water users are often willing to pay for quality reliable water services but are unwilling to pay for unsatisfactory services. As a result, when service level drops due to lack of proper maintenance and responsiveness of the service providers, the water users become less willing to pay which further constraints operations and maintenance activities. They concluded that in most cases, it's not insufficient revenues that leads to failure of rural water points but rather it is mismanagement of collected revenues.

According to Adank & Tuffuor, (2013), the components of sustainable water tariffs for rural water service providers should be able to cover water production costs, routine maintenance and repair works, tariff collection expenses and some extent of spare parts replacement costs This is emphasized by Simukonda et. Al, (2018) who decried the fact that very few water service providers charge tariffs that cover the full costs of supplying water and in developing countries, tariffs are too low even to meet the basic operation and maintenance costs

2.4 Literature review summary and research gaps

From the foregoing review, research and practice of rural water services delivery points out that functionality of rural water points is an outcome of complex systemic interactions between political-economic, technical, financial and environmental factors. While the social norms, political values and economic incentives have an indirect influence on infrastructure functionality since these appear to determine whether the community water users pay for water and how much they pay for it. On the other hand, the practices around management of collected revenues from water use, which includes transparency, determines the effectives of the repair and maintenance activities carried out by the water service providers mandated to manage the water supply scheme giving these a more direct influence on infrastructure functionality. Conversely, reliability of water supply, in the sense of fewer breakdowns interrupting supply, is an outcome of the effectives of the maintenance practices affected which according to the literature review, largely influences water users' willingness to pay and make revenues available for repairs and maintenance.

In terms of the gaps this study seeks to contribute to, much of the literature reviewed indicates substantial work has been done in assessing critical success factors affecting sustainability of rural water points. There're limited studies focusing into detail the specific maintenance and repair activities as well as revenue generation and management practices to ensure funds for maintenance. This study fills in this research gap by focusing on the specific revenue management and maintenance practices adopted by different management models including community management since these have a direct influence over infrastructure functionality in comparison to social norms and political economy factors which have an indirect effect on infrastructure functionality.

2.5 Conceptual framework and definition of study variables

Drawing from the literature, sustainability of rural water schemes is a multi-faced complex interaction of different system elements ranging from social factors, financial, environmental and technical drivers. Functionality is used in this study as a proxy indicator of sustainability. The dependent variable for this study is thus functionality created by assigning 1 to a water source found functional at the time of the study and had not break down for more than 24 hours not more than three times during the preceding two-week period, otherwise it is assigned 0 (non-functional water source). It emerges from the reviewed literature that fundamental to sustaining infrastructure functionality longer is the capacity and effectiveness of carrying out operations and maintenance (O&M) of the facilities. Availability of revenues, how the collected revenues are managed, water user's perceptions of affordability and willingness to pay are factors that facilitate the maintenance practices. For this study, practices on infrastructure maintenance and repairs was observed as the key independent variable. Generation of revenues needed to facilitate the maintenance activities was also observed but only as a factor facilitating the maintenance activities. This relationship is depicted in the conceptual framework shown in figure 3. Based on this conceptual framework, data collections tools were prepared and applied to collect primary data from the sampled rural water supply schemes in Turkana county.



Figure 2-1 Study conceptual framework

Chapter 3 : RESEARCH METHODOLOGY

3.1 The research design and rationale

Quantitative research approaches were primarily used to collect data from water users (household) respondents in this thesis. Focused group discussions with water management committee members as well as key informant interviews with county government and non-governmental agencies was then applied as a follow up to validate and interpreted the quantitative data. Quantitative approach was deemed most appropriate because the type of data on the study variables consisted of largely numerically quantifiable data. Applying the qualitative focused group discussions and key informant interviews allowed the researcher to make interpretations from the quantified statistical results and interpret the data from a knowledge, attitudes and practices perception.

3.2 Sampling

Turkana county with a population of 1,427,797 was the study population. Turkana was purposively selected based on its low levels of access to safe water services (39% of Turkana residents have access to safe water against a national average of 55%) and high non-functionality rates (33% of installed infrastructure are non-functional). Turkana county located in the Arid Semi-Arid Lands (ASALS) of Kenya is also categorized as the poorest county in Kenya with roughly 90% of the residents living below poverty level (Mwangi, 2013).

3.2.1 Sampling frame and procedure

The sampling frame was a list of 41 villages distributed county-wide with water facilities installed in the last 5 years by the county government of Turkana and other Non-Governmental Organisations operational in the county. A simple two-stage cluster sampling procedure was used to select the study respondents. The primary sampling unit, the clusters, were the 41 villages. From this sampling frame, given that the villages have differing population sizes, probability proportion to size sampling was applied so that the larger clusters were given a greater probability of selection. For the follow up key

informant interviews and focused group discussions, purposively selected respondents given their role in rural water services delivery were relied upon.

3.2.2 Sample size

Since the population was finite, the sample size, the number of participants in the study (Frey, 2018), were selected at 95% confidence level such that the margin of error (desired absolute precision) was 5% using the Cochran formulae applying a finite population correction factor as shown in equation 1 (Cochran, 1977; Berenson, Levine, & Szabat, 2015; Lavrakas, 2011). The sample size was computed as Ten (10) sample sites with a total of 201 respondents from the 41 clusters using probability proportional to size sampling calculations. Table 3-1 summarises the sampling procedure used during the study. Table 3-2 summarises the 10 sampled sites at the time when the study proposal wa submitted. Due to unforeseen security challenges in accessing some of the sampled sites during field work, a few of the sampling sites had to be changed. The final sampled sites visited during the study are show in Table 3-3.

$$n_{trt} = \frac{N\hat{p}_{trt}\hat{q}_{trt}}{\frac{d^2}{1.96^2}(N-1) + \hat{p}_{trt}\hat{q}_{trt}}$$

where
 $n_{trt} = \text{sample size}$
 $N = \text{population size}$
 $\hat{p}_{trt} = \text{the estimated proportion}$
 $\hat{q}_{trt} = 1 - \hat{p}_{trt}$
 $d = \text{desired absolute precision}$

Equation 1: Sample size calculation formulae

	Quantitative data (Administer Household	Size
	survey questionnaires)	
Study population	Turkana county	1,427,797
Sampling frame	41 villages where water systems were	140,450
	installed in the last 5 years	
Primary sampling unit	Sampling units were selected from the	10 villages
	sampling frame of 41 villages (clusters)	with total
	using probability proportional to size (PPS),	population
	taking the population of each village as the	
	reference value.	104,108
Secondary Sampling	Within each selected sampling unit (village),	
units	random walk methodology was used to select	
	the respondent households.	
Respondents	An adult member of the household willing to	219
	take part in the survey will be interviewed by	
	a research assistant.	

Table 3-1 Summary of the sampling procedure

Table 3-2:The 10 Initially selected primary sampling units and sizes

	Village Name/Sample site for data collection	Type of water point	population size	sample size
	Kakuma –			
1	Karbokorit	solar/diesel hybrid pump	17,000	33
2	Kataboi	Solar powered pump	6,703	13
	Lodwar town			
	(Kanamkemer,			5
3	Nakwamekwi)	solar/electricity hybrid pump	21,756	42

4	Loperot	Solar powered pump	8,726	17
		solar powered/diesel/electricity		
5	Lokichogio	system	33,700	65
6	Meyan	Solar powered system	1,800	3
7	Naotin	Solar powered system	500	1
		Hand Pump/Solar powered		
8	Kaikor	system	8,700	17
9	Kakelea	Hand Pump	4,028	8
10	Nabulkok	bulkok Hand Pump		2
				201

Table 3-3: The 10 finally selected primary sampling units during field work

	Village Name/Sample site		population	
	for data collection	Type of water point	size	sample size
1	Kaaleng/Kaikor	Hand pump	8,700	24
2	Kakelae	Kakelae Hand pump 402		8
3	Kakuma/Kabokorit	Solar/diesel hybrid system	17000	33
4	Kalokol	Solar powered	470	6
5	Kanamkemer- Lodwar town	er- Lodwar town Solar/electric hybrid 11756		24
6	Kataboi	Solar powered 6703		13
7	Lokichar Hand pump		1227	13
8	Lokichoggio	Electric/diesel hybrid	33700	53
9	Loperot	Solar powered	8726	17
10	Nakwamekwi-Lodwar town	Solar/electric	11878	27
		-		218

3.3 Data collection tools

Household questionnaires was used as the primary data collection tool to collect data from the 219 study respondents. Focused group discussions and key informant interviews were used at a second stage to validate the collected quantitative data from households.

3.4 Data analysis

Given that the data was largely quantitative, various statistical methods including descriptive statistics, frequency counts and regression analysis were applied to analyse the data and process it into useful information for answering the research questions

3.5 Research Quality - validity, reliability and objectivity of the research.

The data collection tools were piloted before the study to ensure that the respondents all understand the question in the same way, so as to ensure both reliability and validity. The pilot was done with 27 household respondents in Nakwamekwi community. Piloting allowed for adjustment in areas where weaknesses of the tools were noted. A key issue that emerged during the pilot was the difficulty the enumerators faced in translating some of the more technical questions into the local Turkana language. The tools were revised to make it easier for the enumerators. The study additionally triangulated the information provided by households with those provided by water management committee members, county government and NGO officials to enhance research quality.

3.6 Ethical considerations during the study

This study applied the following ethical research guidelines to mitigate against any form of harm possible to those involved in the study;

- 1. Consent was given voluntarily, and the participants were informed that consent could be withdrawn at any time during the study.
- The respondents were not required to give their name on any of the questionnaires and strict confidentiality of respondent's identity was maintained.
- Respondents were informed of the duration of the questionnaire to allow then to plan their schedules accordingly.

Chapter 4 : PRESENTATION OF RESEARCH FINDINGS

4.1 Data Collection

Data collection was carried out between December 10th and 21st 2018 at the 10 sampled rural water systems within Turkana county each with different characteristics. Some had solar pumping systems, some diesel generator pumping systems, some hybrid solar-diesel pumping systems and some with handpumps.

Quantitative questionnaires were administered at household level to 218 respondents (water users), both men and women sampled from the 10 communities. The respondents were distributed proportionately across the 10 water systems sampled using probability proportional to size sampling process. Follow up Focused Group Discussions and key Informant Interviews were conducted with water management committees members for each system (averagely 5 committee members per system), 6 county government officials and 4 NGO staff to triangulate and validate the largely quantitative data from water users. Ethical clearance for the study was obtained from Strathmore University Institutional Ethics Review Committee protocol ID SU-IERC0278/18 on the 4th of December 2018.

4.2 Response rate

The study had 10 study sites sampled within the county. Questionnaires were successfully administered to 218 household respondents through face-to face interviews. The response rate recorded was 100%. Figure 4-1 shows the gender of the respondents.



Figure 4-1 Gender distribution of the study respondents

22

4.3 Objective 1 results: Maintenance practices of water service providers

The first research question sought to establish the specific types of infrastructure maintenance activities practiced by rural water service providers in Turkana county. This required a descriptive presentation of the current practices with regard to typologies of repairs and maintenance, response to emerging repair needs and access to spare parts for rural water systems in Turkana county.

4.3.1 Typologies and frequency of maintenance activities

The routine and periodic operations and maintenance and repair activities carried out by the managers of the water points is critical for their sustained operations. A key focus of this study was to understand from the service providers and the water users, the different types of repair and maintenance activities including their frequency and whose responsible. While some requiring less expertise like general cleaning of water point areas are done more regularly, there are more technical activities requiring external intervention which comes far between as shown in table 4-1

Change of oil filters and fuel filters for water systems using a generator pumping set was observed as the only preventive maintenance activity carried out by the service providers. Otherwise majority of the maintenance activity are done only when the need arises like when a pipeline is broken. For the water systems having solar pumping system, only 10% reported cleaning the solar panels at least once per week. Only 2% of the respondents indicated cleaning the livestock watering troughs. 100% indicates that the service providers mentioned carrying out the said maintenance activity with the corresponding frequency

Table 4-1 Typologies and frequencies of maintenance activities by service providers in Turkana

Type of maintenance activity	Frequency done (Done at least)					
	Daily	weekly	monthly	Every 3 months	>3 months	Variable, when need arises
Changing of oil filter						
Changing of fuel filter						
Cleaning of solar panels		生活成				
Cleaning pump area						
Repairing broken pipes		8.85 C - 54				- International
Repairing leaking tanks						
Replacing taps at water kiosks						
Cleaning of livestock watering troughs				NIK SA		
Repair of broken-down hand-pump system (e.g. replacing seals)						
Repair of faulty solar pumping system (e.g. replacement of burnt out control panel)						
Repair of non-functioning water pump inside the borehole						
Repair of broken kiosks and watering troughs						
Repairing of pump replacement of pump parts						

4.3.2 Time taken to repair broken down water facilities

Frequent breaking of pipes and water taps at the water kiosks were listed as the leading cause of water system failures at 57.82% by the respondents as Figure 4-2 shows. Drying up of wells or significant reduction of yield from the water sources leading to insufficient supply of water for both livestock and people came as a third cause of breakdown after pump failures.


Figure 4-2: Causes of water facilities breakdown in Turkana

The water users and water management committee members were asked how long it takes to respond to a facility breakdown when it occurs. While the time it takes to respond was varied across the water points with the majority taking two weeks or more and none completed within a day, the most common response was that the time to repair a broken system depends on the availability of funds. 69% of the water users are of the view that it takes as much time to restore functionality as it takes to make the funds available. This is an interesting response since it was observed that it could be a day or 3 months in some cases. These results are shown in figure 4-3. This trend points towards a lack of a pool of easily available cash flow to respond to repair needs.



Figure 4-3: Time taken to respond to water facility breakdowns in Turkana

While 100% of the committees reported receiving at least five days training on basic operations and maintenance of the water systems mostly from Non-Governmental Organizations (NGOs) and local sub-county water officer who installed the systems, maintenance practices were found to be carried out erratically and response to break downs varied from 3 days to more than three months depending on the nature of work needed and availability of funds. In one community, it took one and half months to acquire and replace a broken seal on a handpump while in another solar powered system, it took 3 months to replace a pump controller that broke down. When asked whether the water facility has enough human resource capacity for maintenance activities, 100% of the responded indicated not. This is because they still have to rely on the sub-county water offices or NGOs sending repair technicians. If both these channels delay, the community collectively contributes and pays a private technician from the nearest urban center i.e. Lodwar town, Kakuma or Lokichogio to come and do the repair. In some cases where the community has a solar powered system, the respondents indicated reaching out to the NGO who installed the system who then facilitates an electrician from Nairobi to repairlike the case of the broken pump controller system at Kaaleng'.

A follow up question on responding to repairs was related to perceptions on who should be carrying out the response to breakdowns. Figure 4-4 indicates that about 50% of the respondents believe that it's the duty of the water service providers, water management committees in this case, to ensure there's enough funds and expertise to carry out the required repairs. About 22% are of the opinion the county government should do it while about 15% interestingly believe it's the water users' responsibility to do the repairs. This is an interesting finding since the water committees do not have sufficient technical skills to carry out the repairs hence they rely on NGOs and county government support.



Figure 4-4: Perceptions on who should carry out water facilities repair in Turkana

4.3.3 Access to spare parts

None of the service providers reported keeping a set of spare parts within their stock. They all purchase from the nearest source when the need arises. The service providers were asked how difficult it was for them to access the spare parts needed for operations and maintenance. The majority, 76.4%, as shown in figure 4-5, felt they have difficulties in access ing spare parts while none felt any ease in accessing spare parts. While some spare parts such as plumbing fittings were available in local hardware stores in local town Centres such as Lokichar, Kakuma and Lokichogio, the more advance spare such as pump parts and electrical components could only be sourced from Lodwar, Kitale and Eldoret where service providers such as Davis & Shirtliff have workshops.



Figure 4-5: Ease of access to spare parts in Turkana

Figure 4-6 shows the responses received when respondents were probed further how they obtain spare parts needed for maintenance and repairs. The major providers of spare parts for the service providers at 53.2%, are NGOs who support the development of community water supplies. It's worth noting that the NGOs provided spare parts support for both water points they developed as well as for water points developed by the government. These findings suggest that distance is a key barrier to accessing spare parts on time and thus quick response to break downs. The furthest study site, Lokichogio is about 214km from Lodwar while Kitale to Lodwar is about 300km.



Figure 4-6: How service providers in Turkana obtain spare parts

4.3.4 Availability and access to post-construction technical support and monitoring

The data shows that the two key external partners to the communities in Turkana in managing their water systems are the county government and NGOs with a specific mention of the Catholic Diocese of Lodwar water program. Majority of the service providers indicated they have received some kind of post construction support-supply of spare parts, support in responding to a breakdown mostly from the NGO, Catholic diocese of Lodwar and periodically from the county government of Turkana. The Catholic diocese of Lodwar water program received a constant mention as a key source of post-construction technical support at all the sampled sites. The NGOs were observed as the only source of any structured technical training and capacity building to members of the water committees for operations and maintenance of the water systems. This is shown in figure 4-7.



Figure 4-7:Sources and kinds of post-construction support in Turkana

As shown in figure 4-8, from the key informant interviews with county government officials and NGO staff, the development of new infrastructure and development of policies was indicated as the key role of the county government in rural water

management. They didn't consider support on regular operations and maintenance as a key role apart from responding to large repair needs that the water management committees haven't been able to respond to on time.



Figure 4-8: Perceptions on the role of county government of Turkana in rural water supply delivery

Limited funding occasioned by adequate county budgetary allocation, weak coordination with NGO actors and conflicts in roles between the county government and the regional water service boards emerged as the key challenges the county government face in their executing their expected direct oversight and monitoring of the activities of rural water service providers on revenue management and technical support as shown in figure 4-9. There are certain community water projects implemented by the national government and NGOs within the county that doesn't involve the local county offices leading to a lack of common understanding and communication on subsequent support mechanisms. This was cited as a significant cause of poor post-construction monitoring and technical support.



Figure 4-9: Challenges in technical support and monitoring of rural water service delivery in Turkana

4.4 Objective 2 results: Revenue management practices of service providers

The second research question sough to establish the specific ways rural water service providers in Turkana county raise are and manage revenues needed for the daily maintenance of the water schemes. From the key informant discussions with the water management committees and county officials, it was observable that 80% (8 out 10) of the water management committees were formed only after the construction of the facilities were completed and handed over to the community. Most committees were non-existence before the project started. The reason for formation of the water committees was so that, on behalf of the communities, they would collect water use tariffs, manage these to carry out on-going maintenance and repairs.

4.4.1 Availability of funds for maintenance: Revenue mobilization means and sources

When asked the different sources of funds used for operations of the system such as buying fuel, repairing broken systems, buying spare parts, as shown in figure 4-10, 87.35% of the committees indicated they collected revenue from water users to help in running the O&M of the water points. The other three sources mentioned were support by NGOs especially during breakdowns to buy spares, support from the county

government local sub-county water engineer, religious organization particularly the catholic church Diocese of Lodwar, community collections (*harambees*) especially when there's a breakdown and the system has been down for a while and in a few cases support by private individual well-wishers. It was interesting to note that the respondents ranked support from NGOs above support from the county government.





4.4.2 Affordability and Willingness to pay for water by water users

When the households were asked whether they pay for water, 70.78% indicated they pay regularly. Of the 70.78%, 54.84% pay at the collection point i.e. kiosk per container as they fetch the water (mostly 20-liter plastic jerrycan) while 41.29% pay a fixed monthly contribution as agreed by the community while the rest, only 3.87%, pay ad hoc when asked to by the water management committee.

When asked whether they find the water tariffs affordable, 59.35 % responded that what they currently pay is too expensive compared to the kind of income they make while 40.65% concurred that the tariffs are reasonably affordable. Table 4-2 shows the average cost of water at the different sample sites. The results indicate lack of a regulated system of charging for water for the rural households in Turkana.

S.	Study Location	Average cost of a 20ltrs				
No		container of water for domestic				
		use (Ksh.)				
1	Kaaleng/Kaikor	3				
2	Kakelae	8				
3	Kakuma/Kabokorit	6.5				
4	Kalokol	5				
5	Kanamkemer	11				
6	Kataboi	5				
7	Lokichar	9				
8	Lokichoggio	5				
9	Loperot	6				
10	Nakwamekwi-Lodwar town	13				

Table 4-2: Average cost of water in various communities in Turkana

In further probing, the water users' who expressed a willingness to pay indicated that the affordability and the reliability of the water points, in terms of it being functional when they need water, has the greatest influence on their willingness to pay for water when asked to. Figure 4-11 shows the factors influencing the water users' willingness to pay for water.



Figure 4-11: Factors influencing water users' willingness to pay for water in Turkana

Revenue collected from the water users at the various water points is used mostly for buying fuel, spare parts, pay staff and conduct annual maintenance of the water points as shown in figure 4-12.



Figure 4-12: How revenues are used by rural water service providers in Turkana

4.4.3 Sufficiency of collected revenues to cover maintenance costs

Only 12.5% of the committee members indicated that the revenue they collect is enough to sustain the water point. 87.5% felt that the revenue collected is not enough to cater for the maintenance of the water points. This means that there's mostly a deficit that is required to manage the water points smoothly. This forces the committee members to seek for money from NGOs, the government and urging community members to make special contributions besides the water tariffs.

4.5 Objective 3 results: Effects of service provider practices on Functionality

The third research question sought to establish the relations between functionality, If any, with the study independent variables. The results from analysis of the maintenance and revenue management practices shows a close interaction between the independent

variables. From the analysis, it emerges that the time taken to repair broken down water facilities is the main predictor variable of functionality. Access to spare parts, access to technical support and monitoring of service delivery, sufficiency of the revenues mobilized to cover operations and maintenance costs all affect the time it takes to repair a broken water system making it endogenous.

Given that the predictor variables are largely categorical, a generalized linear model (GLM) would have been chosen for the analysis. However, the outcome variable, functionality is continuous and there is evidence of endogeneity in the predictor variables. I therefore settled on an instrumental variable regression model to address possible multicollinearity between the predictor variables.

Ex-ante, time taken to repair a broken-down water facility is determined by the type of repairs required, who's responsible for carrying out the maintenance (whether it can be done by water committee members, local technician or whether a technician must be sourced from Kitale) and whether households are required to contribute towards repairing a breakdown when it occurs. These factors are all determinants of functionality but only to the extent they are factors of time taken to repair a broken facility and restore functionality. Hence, I ran an instrumental variable regression model with these factors as instruments, while affordability as an independent predictor in the model as shown in Figure 4-13.

Instrumental v	Numbe Wald Prob R-squ	r of obs chi2(2) > chi2 ared	11 11 11 11 11 11 11 11 11 11 11 11 11	218 15.77 0.0004				
GMM weight mat	rix: Robust			Root	MSE	=	1.4754	
functional-y	Coef.	Robust Std. Err.	z	P> z	[95% 0	Conf.	Interval]	
repair_time afford2 _cons	3695879 .5383256 2.514775	.1719429 .1948343 .7947623	-2.15 2.76 3.16	0.032 0.006 0.002	70658 .15645 .95706	97 575 593	032586 .9201937 4.07248	
Instrumented: Instruments:	repair_time afford2 1.re	epairs 2.rep	airs 4.r	epairs 1.	who_repai	rs		

2.who_repairs 3.who_repairs 4.who_repairs 5.who_repairs hh_contribute

Figure 4-13: Regression analysis results

To test the fitness of the model, I checked endogeneity to confirm that the exclusion criterion is met. I also checked that there's indeed correlation among the variables included at the first stage regression. Figures 4-14 and 4-15 shows the results.

```
. estat endogenous
Test of endogeneity (orthogonality conditions)
Ho: variables are exogenous
GMM C statistic chi2(1) = 5.0582 (p = 0.0245)
```

Figure 4-14: Results of testing the fitness of the regression model

First-stage regression summary statistics

Variable	R-sq.	Adjusted R-sq.	Partial R-sq.	Robust F(9,207)	Prob > F
repair_time	0.0732	0.0284	0.0726	2.61849	0.0069

Figure 4-15: Results of checking for correlation among predictor variables

The endogeneity test shown in figure 4-14 indicates a significant p-value of 0.0245 at 95% confidence level. The null hypothesis that the instrumented variable, time taken to repair a broken facility, is exogenous justifying the choice of an instrumental variable

regression. On the other hand, figure 4-15 shows that the F-Statistic in the first stage regression test indicates a significant value of 0.0069 at 95% level. I therefor conclude that there is significant correlation between the variables included in the first stage regression and the instrumented variable, time taken to repair a broken-down water facility.

Summary and interpretation of regression coefficients

A further summary of the regression coefficients is shown in the table 4-3. In summary, the results show that a water system is likely to be considered functional if it takes (0.37 or 37%) less time to repair compared to the base category of less than 1 day (24 hours).

Variables	functionality
func_repair_time	-0.370**
	(0.172)
afford2	0.538***
	(0.195)
Constant	2.515***
	(0.795)
Observations	218

Table 4-3: Summary of regression results

Another factor that determines functionality is affordability of the water by the households. A water point would be functional for a longer time if households could afford the water by 54% points (price reduction recommended here to increase affordability). The results show that there exists a cyclic relationship in which affordability increases access which in turn improves functionality as more households contribute towards maintenance, hence reducing repair time.

Chapter 5 : DISCUSSION OF RESULTS

5.1 Introduction

The purpose of this study was to assess the effects of infrastructure maintenance practices of the rural water service providers (in this study mostly community-level water management committees) in Turkana county on the facilities functionality. The literature review pointed that fundamental to sustaining rural water supply infrastructure functionality is the actual maintenance practices as well as capacity and effectiveness of capturing and efficiently utilizing resources for operations and maintenance. This chapter presents a discussion of the study findings as related to the literature on maintenance practices in water services delivery as well, revenue management practices as well as on rural water sustainability and functionality as presented in chapter tow of this thesis.

The study sought to answer three research questions: (R1) What are the types of maintenance activities practiced by rural water service providers in Turkana county, (R2) How do rural water service providers in Turkana county raise and manage revenues needed for the maintenance of the schemes and lastly, (R3) How do the maintenance and revenue management practices by rural water service providers affect the functionality of the water schemes.

5.2 Summary discussions of the study findings

Majority of the maintenance activities done by the water service providers in Turkana county were only done as a reactive/curative measure after the system has broken down. There's little evidence of any scheduled preventive maintenance measures save for simple daily routine tasks such as cleaning of water access points. When a break down occurs at a water system, the time taken to respond is varied with the most common response, 69%, being that it "takes as long as money is available". Only 0.2% of breakdowns are repaired within 24 hours. In some cases, depending on the magnitude of repairs needed and access to spare parts, the response time has taken as much as 3 months to restore functionality after a breakdown. On revenue management practices, the results indicated that majority of the funds for operating and running the water systems, about 87%, come from the water

users as water tariffs or from periodic collective contributions, locally called *Harambees*. The results indicate that the communities periodically reached out to NGOs or the county government when the need for a major repair arises.

While the results reveal a high willingness to pay for water at 71% of the water users, more than half of the respondents indicated that the current amounts they were paying for water is too high. Almost half, 43% of the revenues collected are used to buy fuel for water points using diesel pumping sets. The bulk of the revenues, 57% is used to buy spare parts, pay operations staff and to carry out regular maintenance such as annual overhaul of diesel generators. 88% of the water service providers indicated that the revenues collected are not sufficient to cover their operations and maintenance needs.

The study findings revealed a close interaction between the independent variables themselves with the time taken to repair broken down water facilities emerging as the main predictor variable of functionality. The other predictor variable; access to spare parts, access to technical support and monitoring of service delivery, sufficiency of the revenues mobilized to cover operations and maintenance costs all affect the time it takes to repair a broken water system making it an endogenous variable. This necessitated an instrumental variable regression analysis to establish relations between the predictor variables and functionality status of water points. The regression results indicate that a rural water system in Turkana is likely to be functional if it takes (0.37 or 37%) less time to repair compared to the base category of less than 24 hours. Affordability of water by the households emerged as another key factor that determines functionality since it influences sufficiency of revenues needed to respond to a break down. The regression results indicate that a water point in Turkana would be functional for a longer time if households could afford the water by 54 % points (calls for a reduction of the current water tariffs by 54%). The results demonstrate a cyclic relationship in which affordability increases access which in turn improves functionality as more households contribute towards maintenance, hence reducing repair time.

5.3 Analysis of the study findings within the context of other literature

The results from this study confirm many of the concerns around sustainability of rural water supply schemes and the inefficiencies common with the community-based management model as highlighted in chapter 2. In chapter two, studies by Leclert, Nzioki, & Feuerstein (2016); Walters & Javernick-Will (2015); Skinner (2009) and Rural Water Supply Network (2010) indicated that the underlying causes of high non-functionality rates of rural water facilities lies in the ineffectiveness of the maintenance practices and revenue management practices. The findings of this study confirm these.

The results of this study indicate that the expectations of the community management model adopting a service delivery approach as proposed by Lockwood et. Al, (2010) are difficult to achieve for remote water point systems like those in Turkana. This inefficiency and ineffectiveness in service delivery is largely affected by a complex interaction of factors including insufficient self-capacity to carry out effective preventive maintenance, difficulties in accessing spare parts within the shortest time possible, constraints in getting skilled technical support for large repair works and well-plane monitoring by county government, insufficiency of collected funds from water users to cover the maintenance needs in its entirety, all which directly influence the time it takes to repair a water system when it becomes non-functional. These findings, that the different independent variables affecting ultimate functionality of water points are intricately inter-related with each other confirms the findings of Peter & Nkambule (2012); Enéas da Silva, et al., (2013); Jones, et al., (2013), Jones, Anya, Stacey, & Weir, (2012), Amjad et al., (2015) that a balance between then availability of skilled technicians, access to high quality spare parts as well as generation of sufficient revenues to cover the cost of maintenance and repairs contribute to functionality of rural water points. The water committee members in Turkana indicated that the time it takes to repair a broken system depends on availability of funds needed for the repair. On the other hand, water users' willingness to pay for water, which makes funds available, is directly affected by their perception on affordability and how reliable the water supply is. This shows how strong interactions between technical factors and financial factors affect functionality. The results concur with Walters & Javernick-Will (2015) assertion that dynamic and systemic interactions of technical, social, financial, institutional, and environmental factors often lead to premature water system failure.

The literature asserts that the performance of water supply infrastructure depends on the maintenance regime adopted (Simukonda, Farmani, & Butler, 2018). The findings of this study pointed out that the service providers in Turkana do not have any structured preventive maintenance program instead often responding to breakdowns only when they occur. The intermittent functionality, where it takes as much as 3 months to respond to a breakdown can be attributed to a failure of the service providers to adopt a proper maintenance regime

While Kivuva (2014) observed that the sustainability of community water projects in Kutui county was attributed to the acquisition of management skills, technical operation and maintenance skills as well as acquisition of resource mobilization skills by water uses committees highly contributed, the findings from Turkana county indicates that despite all the water management committees managing the water points received training and capacity building on operations, maintenance and financial management, their effectiveness in responding to breakdowns is still low with some repairs taking as long as three months, funds are still often collected from water users when the need for a major repair arises instead of drawing from a well-managed savings account and they have no properly well planned maintenance regime. The results of this study corroborate Chown's (2014) conclusions that the community-management model is characterized by neglect of maintenance, slow and substandard repairs, and failure of committees to save sufficient funds.

The results on factors influencing the water users' in Turkana willingness to pay, highlighting reliability in terms of the system functioning when they need water, agrees with by Koehler, Thomson, & Hope (2015) who observed that payments made by water users in rural areas is highly continent on levels of service provided besides the affordability question.

Shah & Mills (2018) posited that even the poorest water users are often willing to pay for quality reliable water services but are unwilling to pay for unsatisfactory services. As a result, when service level drops due to lack of proper maintenance and responsiveness of

the service providers, the water users become less willing to pay which further constraints operations and maintenance activities. The finding of this study showing the existence of a cyclic relationship in which affordability increases access which in turn improves functionality as more households contribute towards maintenance, hence reducing repair time strongly agrees with Shah & Mills position.

Chapter 6 : CONCLUSIONS AND RECOMMENDATIONS

6.1 Conclusions

This chapter presents the conclusions that can be drawn from the results, presents significance of this study's findings and makes recommendations for action to policy makers. Also highlighted are suggestions for further research.

From the presented study findings and the discussions in chapters 4 and 5, the following conclusions are made:

There's a tendency of the government and other actors such as NGOs of focusing mostly on the short-term new water infrastructure development for rural populations with little focus given to operations, maintenance and monitoring systems strengthening support post-construction. This negatively impacts on the long-term functionality of the facilities since communities with little technical capacity and limited financial resources are expected to manage these facilities.

Rural water service delivery in Turkana is done without a clear institutional strategy where service providers are not formally held to account on compliance or noncompliance with government service delivery standards. By virtue of the service providers under the community-management model operating in Turkana being "voluntary" and not a formally registered and regulated service provider, it is not possible to hold them to account on the quality and effectiveness of the services they offer. This is a significant policy challenge for the effective delivery of an essential public service like water.

The persistent non-functionality and frequent failure of rural water systems is a result of a cyclic relationship between poor response by the service providers leading to unwillingness to pay by water users which led to lack of adequate funds for maintenance and response to breakdowns.

Affordability of water remains a key issue for rural water users. When the water cost is too high, as most users have lamented, there is no sufficient generation of revenues to cover the cost of operations and maintenance since users can't pay up. In addition to this, there's no uniformity in water tariffs for the rural households with some household paying

Ksh. 3 while others pay Ksh. 11 for the same volume of water. This points to a significant policy gap on setting of rural water tariffs.

The prevalent "capacity building" approach for rural water service providers involve providing short-term, 5-10 days one-off training in a bid to help them acquire management skills, technical operation, maintenance skills, resource management skills. This is done without a well-organized post-constriction technical support and monitoring regime by the local government. Such a state is not likely to contribute to increased functionality of the water systems as was evident in Turkana. This is underpinned by two factors, one that the members of these committees are often village elders not well educated so their capacity to acquire the kind of knowledge expected of them is low and secondly that no matter how well trained and skilled the committee members are, they most often operate on a "voluntary" basis thus difficult to enforce regulatory standards for water services provision.

The community management model where members of the beneficiary community appoint or elected a few of them to members of a "voluntary" committee responsible for directly operating and managing the water point has proved non-responsive and ineffective in ensuring the sustained functionality of the systems. Significant changes are required in formulating the structures of how rural water services are delivered including possible models where the communities still retain the powers to oversee the management of the services but are not directly responsible for day to day operations and maintenance of the system.

6.2 Policy significance of the study findings

Sustaining rural water supply post-construction remains a significant policy issue for both the government and development assistance organizations. The averagely 30-45% rural water facilities failure rate post-construction is unacceptable given the high capital investment made into these facilities. Several approaches and models have been proposed and tested in attempts to address this policy challenge. Among these include the predominant community-based management model as well as testing private sector-led models. These have brought about mixed results. As this study points in corroboration

with other studies, the community-based management model has yielded much less than expected success with different community-led water service providers characterized by neglect of maintenance, slow and substandard repairs, and failure of committees to save sufficient funds. Attempts to bring in private sector service providers has been resisted with significant value-questions as to the motives of private sector profits orientation in supplying a public good as well as concerns around commercial viability of remote rural water points to attract private market players.

The findings of this study which casts a grim picture of the current service delivery models and practices in rural water supply calls for significant paradigm shift towards a wholistic system-view of the policy issues in rural water services delivery. Governments and development agencies need to think "outside the box' to find a sustainable solution. The policy significance of this study's findings is to call for a greater "duty of care" approach and designing rural water service provision with an imbedded sustainability from the onset. Most often, as the study has pointed out, low capacity community water users are often left alone to manage a water supply system with very minimal training for a demanding task. They are often left with little structured institutional and technical support to ensure the facilities function reliably. The government and NGOs unfortunately later on blame 'poor management by communities" on the failure of rural water facilities. Duty of care as a policy approach demands that the financiers and developers of the infrastructure establish a forward-looking system and makes explicit provisions for how the technical maintenance of the systems will be done 5 or 10 years down the line and how these will be financed. Imbedded sustainability as a policy approach calls for a well-structured maintenance regime that goes beyond curative "quick fixes" to empowering local community service providers with a technical preventive maintenance support model by the county government and less dependent on external agencies such as NGOs. This calls for local governments such as Turkana county government to review their water sector financing strategy towards a system costing strategy which allocates fund for both capital infrastructure development, subsequent operations and maintenance as well as strengthening the enabling environment for sustainability of services. A financing strategy that is informed by identified and mitigated sources of revenue leakages, leverages on non-traditional funding sources such

as private capital and capitalizes on efficiency gains such as reducing physical and commercial water loses and professional management.

STRATHMORE UNIVERSITY

6.3 Recommendations

Based on the evidence from this study, the following recommendations are made to trigger changes in water management models that will lead to improved functionality of rural water facilities. The proposed recommended solutions revolve around the policy significance of the issue and seeks to establish a duty of care and imbedded sustainability in rural water supply systems.

Turkana county government, as the government responsible water and sanitation services delivery, to adopt a system-wide approval mechanism of new infrastructure development. The project sponsors, whether the government or non-state actors like NGOs, charities should work with the communities and local government support mechanisms to design ex-ante before construction, a system of how the facilities will be maintained after construction imbedded within clear institutional structures that can be held to account on the quality of services provided. Currently the practice is to budget for the capital infrastructure development without any clear mechanisms on how the life-cycle maintenance requirement of the system will be met.

County government of Turkana to adopt a professional service delivery model for the management of rural water schemes over the current "voluntary" community-based model. Two specific models are proposed. One is having county-owned commercial water service providers such as Lodwar Water and Sanitation company Ltd directly manage the rural water points while the local community water users form a consumer association with oversight of the operations of the company. Through this, the community will still be actively engaged in the management of the services as an oversight board while the company directly executes professionally as per regulatory indicators developed by WASREB the operations and maintenance. Secondly, pilot private sector interventions to improve the management, delivery, capacity and operations of service providers through a delegated management model where the county government water department or the commercial water company whose coverage area accompanies rural

water points, contract through a performance-based service contract, a private service provider to manage a single or multiple bundled water points.

Application of a more systematic water tariffs collection and management through automated systems such as pre-paid meters. In this model, the water service provider would issue a water access tokens to registered water users/household, load it with money say Ksh.100 then they access water from the nearest kiosk to his/her dwelling. Such a system promises to seal loopholes for funds leakages which will make more funds available for maintenance and repairs needed.

Regulation of water tariffs charged by service providers to rural households. Currently, the regulation of water tariffs only covers areas served by the commercially registered water companies such as Lodwar water. There is no system for determining the amounts rural households served with point water sources pay per volume of water consumed or fixed monthly charge. The county should enact regulations specifying how much the service providers should charge say per 20 liters of water.

Setting up a county-wide Water schemes maintenance fund to ensure there's enough funds for operations and maintenance of rural water systems. The funds would be pooled from an Operations and maintenance tax ("O & M Tax") out of each new capital infrastructure development. Every new development, whether done by the government or NGOs, would be required to set a percentage of the total capital cost of the project into the maintenance fund for the system's subsequent maintenance. This could be augmented by other funding sources such specific county budget appropriations for maintenance over and above the capital development funds. Social impact investors could also be mobilized to channel their resources into this common pool fund. The funds from this pool would then be disbursed linked to specific performance indicators to service providers who would first put in an application to the fund manager for maintenance support.

6.4 Suggestions for further research

This study's scope was limited to the more "technical" issues of maintenance and revenue management practices of rural service providers. While these are important, they only but a part of a larger context political economy and governance factors. Political factors

driving community quest for justice and economic drivers giving incentives and constraining different individual and collective actors do lay a large part in which management models are chosen and their outcomes. An area for further research proposed is an exploration of the governance processes and political economy factors driving the predominant practices in rural water services delivery how these affect the functionality of rural water schemes.

The author suggests research that will analyze empirical evidence collected through mixed qualitative and quantitative methods on how formal and informal institutional arrangements, stakeholder interactions including incentives and constraints they face, social constructs and cultural norms underpinned by diverse structures affect the management model choices and their effectiveneness on sustaining rural water services provision.

REFERENCES

- Adank, M., & Tuffuor, B. (2013). Management models for the provision of small town and periurban water services in Ghana. Accra: TPP Project/RCN Ghana. WASH Resource Centre Network (RCN) Ghana.
- Adank, M., Kumasi, T. C., Abbey, E., Dickinson, N., Dzansi, P., Atengdem, J. A., ... Laari Chimbar, T. (2013). The status of rural water services in Ghana : a synthesis of findings from 3 districts (Akatsi, Sunyani West and East Gonja Districts). IRC.
- Advocates' Coalition for Development and Environment (ACODE). (2012). Governance Aspects in the Water and Roads Sectors in Uganda. Kampala, Uganda: Advocates' Coalition for Development and Environment.
- Ahmad, J. (2004, December 18). *Making Services Work for Poor People: Water and Sanitation* [Presentation at the OECD meeting in Paris].
- Akiba Mashinani Trust, University of Nairobi, Slum Dwellers International, University of California Berkeley, Strathmore University, & Katiba Institute. (2017). *Mukuru situational analysis*. Slum Dwellers International.
- Alexander, K. T., Tesfaye, Y., Dreibelbis, R., Abaire, B., & Freeman, M. C. (2015). Governance and functionality of community water schemes in rural Ethiopia. *International Journal of Public Health*, 60(8), 977-986.
- Amjad, U., Ojomo, E., Downs, K., Cronk, R., & Bartram, J. (2015). Rethinking Sustainability, Scaling Up, and Enabling Environment: A Framework for Their Implementation in Drinking Water Supply. *Water*, 7(12), 1497-1514. doi:10.3390/w7041497
- Anderson, S. P., & Coate, S. (2000). Market Provision of Public Goods: The Case of Broadcasting. Cambridge, MA: National Bureau of Economic Research.
- Baietti, A., Kingdom, W., & Van Ginneken, M. (2006). Characteristics of wellperforming public water utilities. World Bank Group.
- Bakker, K. (2005). Neoliberalizing Nature? Market Environmentalism in Water Supply in England and Wales. Annals of the Association of American Geographers, 95(3), 542-565.

- Bakker, K. (2010). *Privatizing Water: Governance Failure and the World's Urban Water Crisis.* Ithaca, NY: Cornell University Press.
- Bakker, K. (2014). The Business of Water: Market Environmentalism in the Water Sector. Annual Review of Environment and Resources, 39(1), 469-494.
- Bakker, K. J. (2003). A Political Ecology of Water Privatization. *Studies in Political Economy*, 70(1), 35-58.
- Barlow, M., & Clarke, T. (2012). Blue gold: The battle against corporate theft of the world's water. New York, NY: New Press.
- Batley, R., & Mcloughlin, C. (2015). The Politics of Public Services: A Service Characteristics Approach. World Development, 74, 275-285. doi:10.1016/j.worlddev.2015.05.018
- Beecher, J. A. (2013). What matters to performance? Structural and institutional dimensions of water utility governance. *International Review of Applied Economics*, 27(2), 150-173. doi:10.1080/02692171.2012.752447
- Beecher, J. A. (2013). What matters to performance? Structural and institutional dimensions of water utility governance. *International Review of Applied Economics*, 27(2), 150-173.
- Behailu, B. M., Hukka, J. J., & Katko, T. S. (2016). Service Failures of Rural Water Supply Systems in Ethiopia and Their Policy Implications. *Public Works Management & Policy*, 22(2), 179-196. doi:10.1177/1087724x16656190
- Berenson, M. L., Levine, D. M., & Szabat, K. A. (2015). *Basic business statistics: Concepts and applications* (13th ed.). Pearson.
- Bishop, J. D. (1995). Adam Smith's invisible hand argument. *Journal of Business Ethics*, 14(3), 165-180.
- Boelee, E., Chiramba, T., & Khaka, E. (2011). An ecosystem services approach to water and food security. Colombo: International Water Management Institute.
- Bonsor, H., MacDonald, A., Casey, V., Carter, R., & Wilson, P. (2018). The need for a standard approach to assessing the functionality of rural community water supplies. *Hydrogeology Journal*, 26(2), 367-370.

- Borenstein, Severin. (2008). Principal-agent incentives, excess caution, and market inefficiency: Evidence from utility regulation. Cambridge, MA: National Bureau of Economic Research.
- Boulenouar, J., & Schweitzer, R. (2015). Infrastructure asset management for rural water supply. Hague, Nertherlands: IRC.
- Brinkerhoff, D. W., & Bossert, T. J. (2013). Health governance: principal-agent linkages and health system strengthening. *Health Policy and Planning*, 29(6), 685-693.
- Bruijn, H. D., & Dicke, W. (2006). Strategies for safeguarding public values in liberalized utility sectors. *Public Administration*, 84(3), 717-735.
- Carter, R., Harvey, E., & Casey, V. (2010, November). User financing of rural hand pump water services. Paper presented at IRC Symposium 2010: Pumps, Pipes and Promises, Hague,Nertherlands.
- Castro, A. (2018, May 10). Blended Finance: a key to achieve universal access to water supply and sanitation by 2030 [Web log post]. Retrieved from http://blogs.worldbank.org/water/blended-finance-key-achieve-universal-accesswater-supply-and-sanitation-2030
- Castro, J. E., & Heller, L. (2012). Water and Sanitation Services: Public Policy and Management. London, UK: Routledge.
- Catarina, F., Smits,, S., Nyarko,, K., Naafs, A., & Franceys, R. (2013). Financing capital maintenance of rural water supply systems: current practices and future options. IRC International Water and Sanitation Centre.
- Chepyegon, C., & Kamiya, D. (2018). Challenges Faced by the Kenya Water Sector Management in Improving Water Supply Coverage. *Journal of Water Resource* and Protection, 10(01), 85-105.
- Chowns, E. E. (2014). The political economy of community management: a study of factors influencing sustainability in Malawi's rural water supply sector (Doctoral dissertation). Retrieved from http://etheses.bham.ac.uk/5014/

Cochran, W. G. (1977). Sampling techniques (3rd ed.). New York: J. Wiley.

Consumer federation of Kenya. (2014, March 17). Why the Water Bill, 2014 is generating boiling point temperatures as stakeholders jostle to either retain or drop the "watered down" final draft. Retrieved December 15, 2017, from http://www.cofek.co.ke/index.php/news-and-media/396-why-the-water-bill-2014-is-generating-boiling-point-temperatures-as-stakeholders-jostle-to-either-retain-or-drop-the-watered-down-final-draft?showall=

- Cornes, R., & Sandler, T. (1999). The theory of externalites, public goods, and club goods. Cambridge: Cambridge University Press.
- Cowen, T. (2001). *Public goods and market failures: A critical examination*. New Brunswick: Transaction Publishers.
- D'Arcy, M. (2014, May). State Capacity, Democracy and Public Goods Provision: Theory and Empirical Evidence. Paper presented at The Quality of Government and the Performance of Democracies, Gothenburg.
- Davis, J., Komives, K., Prokopy, L., Thorsten, R., & Whittington, D. (2008). How well is the demand-driven, community management model for rural water supply systems doing?: Evidence from Bolivia, Peru, and Ghana. Manchester: Brooks World Poverty Institute.
- Dewachter, S., Holvoet, N., Kuppens, M., & Molenaers, N. (2018). Beyond the Short versus Long Accountability Route Dichotomy: Using Multi-track Accountability Pathways to Study Performance of Rural Water Services in Uganda. World Development, 102, 158-169. doi:10.1016/j.worlddev.2017.09.018
- Enéas da Silva, F. O., Heikkila, T., De Souza Filho, F. D., & Costa da Silva, D. (2013). Developing sustainable and replicable water supply systems in rural communities in Brazil. *International Journal of Water Resources Development*, 29(4), 622-635. doi:10.1080/07900627.2012.722027
- Environmental Protection Agency. (2017). Effective utility management: A primer for water and wastewater utilities. Washington, D.C.: U.S. Environmental Protection Agency.
- European Commission. (2004). *Handbook of promoting good governance in EC development and co-operation*. Brussels, Belgium: Author.
- European Commission. (2008). Analysing and Addressing Governance in Sector Operations. Brussels, Belgium: Author.
- Fernando, A. C. (2012). Corporate governance: Principles, policies and practices (2nd ed.). New Delhi, India: Pearson Education.

- Fogelberg, K. (2013). From Adopt-a-Project to Permanent Services: The Evolution of Water For People's Approach to Rural Water Supply in Bolivia. Water Alternatives, 6(3), 367-383.
- Fogelberg, K. (2013). Water Alternatives. From Adopt-a-Project to Permanent Services: The Evolution of Water for People's Approach to Rural Water Supply in Bolivia, 6(3), 367-383.
- Fonseca, C., & Pories, L. (2017). Financing WASH: how to increase funds for the sector while reducing inequities: Position paper for the Sanitation and Water for All Finance Ministers Meeting April 19, 2017. IRC.
- Foster, V. (1996). *Policy Issues for the Water and Sanitation Sectors*. Inter-American Development Bank.
- Franks, T. R. (2004). *Water governance what is the consensus?* University of Bradford. Department for International Development.
- Frey, B. B. (2018). The Sage encyclopedia of educational research, measurement, and evaluation (Vols. 1-4th ed.). Thousand Oaks, CA: SAGE Publications Ltd.
- Friedrich-Ebert-Stiftung. (2012). Regional Disparities and Marginalisation in Kenya. Nairobi, Kenya: Author.
- Furlong, K. (2010). Neoliberal Water Management: Trends, Limitations, Reformulations. Environment and Society, 1(1).
- Futurepolicy.org. (2015, February 10). The Remunicipalisation of Water Services in Paris, France. Retrieved December 15, 2017, from http://www.futurepolicy.org/food-and-water/remunicipalisation-of-waterservices-paris/
- Gailmard, S. (2014). Accountability and Principal-Agent Theory. In *The Oxford* Handbook of Public Accountability. Oxford, UK: Oxford University Press.
- Giordano, F., Mussari, R., & Caputo, F. (2014). Managing services and voicing public values: the water sector in Italy. In *Governance and public management: Strategic* foundations for volatile times. New York [u.a.: Routledge.
- Giordano, F., Mussari, R., & Caputo, F. (2014). Managing services and voicing public values. In Governance and public management: Strategic foundations for volatile times (pp. 74-92). London, uk: Routledge.

- Government of Kenya. (2009). Sample Survey on Water Quality and Functionality of Water Systems. Ministry of Water and Irrigation.
- Government of Kenya. (2015). Kenya demographic and health survey, 2014: Key indicators. Kenya National Bureau of Statistics.
- Government of Kenya. (2016, February 22). Speech by His Excellency Hon.Uhuru Kenyatta during the 18th African Water Association International Congress and Exhibition at KICC on 22nd February 2016. Retrieved December 15, 2017, from http://www.president.go.ke/2016/02/22/speech-by-his-excellency-hon-uhuru-kenyatta-c-g-h-president-and-commander-in-chief-of-the-defence-forces-of-the-republic-of-kenya-during-the-18th-african-water-association-afwa-international-con/
- Government of Kenya. (2017). Report of the implementation of the agenda 2030 for Sustainable development goal 6 in Kenya. Ministry of Water and Irrigation.
- Gruber, J. (2011). Public finance and public policy. New York: Worth Publishers.
- Harvey, P. (2011). Sustainable Supply Chains for Rural Water Services. Linking local procurement of hand pumps and spare parts supply (Field Note No 2011-1). Rural Water Supply Network.
- Harvey, P. A., & Reed, R. A. (2007). Community-managed water supplies in Africa: sustainable or dispensable? *Community Development Journal*, 42(3), 365-378.
- Hefetz, A., & Waner, M. (2004). Privatization and Its Reverse: Explaining the Dynamics of the Government Contracting Process. *Journal of Public Administration Research and Theory*, 14(2), 171-190. doi:10.1093/jopart/muh012
- Hellegers, P. J. (2011). *Water: the world's most valuable asset*. Wageningen: Wageningen UR.
- Hepworth, N. (2016, August). Social accountability for a water-secure future: knowledge, practice and priorities. Paper presented at World Water Week,, Stockholm,Sweden.
- Hillman, A. (2009). *Public Finance & Public Policy* (2nd ed.). New York, NY: Cambridge University Press.

- Honkonen, T. (2017). Water Security and Climate Change: The Need for Adaptive Governance. Potchefstroom Electronic Law Journal/Potchefstroomse Elektroniese Regsblad, 20, 1-26.
- Hope, R., Foster, T., Money, A., & Rouse, M. (2012). Harnessing mobile communications innovations for water security. *Global Policy*, 3(4), 433-442.
- Hufty, M. (2011). Investigating Policy Processes: The Governance Analytical Framework (GAF). In Research for Sustainable Development: Foundations, Experiences, and Perspectives (pp. 403-424). Bern, Switzerland: Geographica Bernensia.
- Independent Evaluation Group (IEG). (2012). Sourcebook for Evaluating Global and Regional Partnership Programs: Indicative Principles and Standards. Washington, DC: World Bank.
- International Water and Sanitation Centre. (2012). A service delivery approach to rural water supply. The Hague, Netherlands: IRC.
- IRC International Water and Sanitation Centre. (2012). *Indicators for sustainable rural water supply services in Ghana*. Hague, Nertherlands: Author.
- Jacobson, M., Meyer, F., Oia, I., Reddy, P., & Tropp, H. (2013). User's guide on assessing water governance. United Nations Development Programme.
- James, L., Trémolet, S., & Ikeda, J. (2016). Achieving Universal Access to Water and Sanitation by 2030: The Role of Blended Finance.". Washington: World Bank.
- Jones, S., Anya, A., Stacey, N., & Weir, L. (2012). A Life-cycle Approach to Improve the Sustainability of Rural Water Systems in Resource-Limited Countries. *Challenges*, 3(2), 233-260. doi:10.3390/challe3020233
- Jones, S., Sanford Bernhardt, K., Kennedy, M., Lantz, K., & Holden, T. (2013). Collecting Critical Data to Assess the Sustainability of Rural Infrastructure in Low-Income Countries. Sustainability, 5(11), 4870-4888. doi:10.3390/su5114870
- Julie, I. S. (2014). MARKET MECHANISMS FOR VALUING PUBLIC GOODS (Doctoral dissertation). Retrieved from http://digitalcommons.uri.edu/oa_diss/207

- Kaufmann, D., Aart, K., & Pablo, Z. L. (1999). Aggregating governance indicators (English). Policy, Research working paper. Washington, DC: World Bank.
- Kenya National Bureau of Statistics. (2010). Kenya Population and Housing Census 2009. Author.
- Kjaer, A. M. (2010). Governance. Cambridge, MA: Polity Press.
- Koehler, J., Thomson, P., & Hope, R. (2015). Pump-Priming Payments for Sustainable
 Water Services in Rural Africa. World Development, 74, 397-411.
 doi:10.1016/j.worlddev.2015.05.020
- Kwena, R., & Moronge, M. (2015). Determinants of sustainability of rural water projects in Kenya: a case study of the Netherlands DevelopmentOrganisation (SNV) supported water schemes in Kajiado County. The Strategic Journal of Business and Change Management. Strategic Journal of Business and Change Management, 2(2), 2025 – 2077.
- Laban, P. (2007). Accountability and Rights in Right-based Approaches for Local Water Governance. International Journal of Water Resources Development, 23(2), 355-367.
- Landell-Mills, P., & Serageldin, I. (1991). Governance and the External Factor. *The World Bank Economic Review*, 5(suppl 1), 303-320.
- Langdown, I. (2018). Results of a mid-line survey for the Sustainable Water In Fragile context (SWIFT) programme in Kenya and the DRC. London, UK: Oversea Development Institute (ODI).
- Lavrakas, P. J. (2011). *Encyclopedia of survey research methods*. Thousand Oaks, CA: SAGE Publications.
- Le Strat, A. (2010). Paris: An example of how local authorities can regain control of water management. Reclaiming Public Water Network.
- Leclert, L., Nzioki, R. M., & Feuerstein, L. (2016). Addressing Governance and Management Challenges in Small Water Supply Systems – The Integrity Management Approach in Kenya. *Aquatic Procedia*, 6, 39-50.
- Leruth, L., & Paul, E. (2008). A Principal-Agent Theory Approach to Public Expenditure Management Systems in Developing Countries. OECD Journal on Budgeting, 7(3), 1-29.

- Lockwood, G., & Smits, U. (2011). Supporting Rural Water Supply. Moving towards a Service Delivery Approach. Hague, Nertherlands: International Water and Sanitation Centre (IRC).
- Lockwood, H. (2003). Post-Project Sustainability: Follow-up Support to Communities Literature and Desk Review of RWSS Project Documents. World Bank.
- Lockwood, H. (2014, November 25). Understanding the service delivery approach: the political economy of delivering rural water.
- Lockwood, H. (2014, October 23). Understanding the service delivery approach: the political economy of delivering rural water [PPT].
- Lockwood, H., & Le Gouais, A. (2014). Service delivery indicators and monitoring to improve sustainability of rural water supplies. IRC.
- Lockwood, H., & Smits, S. (2011). Supporting Rural Water Supply: Moving towards a Service Delivery Approach. Rugby, UK: Practical Action Publishing Ltd.
- Lockwood, H., & Smits, S. (2011). Supporting rural water supply: Moving towards a service delivery approach. Warwickshire, UK: Practical Action Publishing Ltd.
- Lockwood, H., Smit, S., Schouten, T., & Moriarty, P. (2010, April). *Providing Sustainable Water Services at Scale*. Paper presented at International Symposium on Rural Water Services, Kampala.
- Lockwood, H., Smits, S., Schouten, T., & Moriarty, P. (2010, April). *Providing Sustainable Water Services at Scale*. Paper presented at International Symposium on Rural Water Services, Kampala, Uganda.
- Makris, M. (2003). The Theory of Incentives: The Principal-Agent Model. *The Economic Journal*, *113*(488), F394-F395.
- Martinez, S. P. (2014). Integrated water resources management in the 21st century: revisiting the paradigm. Boca Raton, FL: CRC Press, Taylor & Francis Group.
- McCord, P., Dell'Angelo, J., Baldwin, E., & Evans, T. (2016). Polycentric Transformation in Kenyan Water Governance: A Dynamic Analysis of Institutional and Social-Ecological Change. *Policy Studies Journal*, 45(4), 633-658. doi:10.1111/psj.12168

- Mekonta, I., Butterworth, J. A., & Holslag, H. (2015, July). Great expectations: Selfsupply as a formal service delivery model for rural water in Ethiopia. Paper presented at 38th WEDC International Conference, Loughborough, UK.
- Mekonta, L., Butterworth, J., & Holtslag, H. (2015). Self-supply as a formal service delivery model for rural water in Ethiopia : ambition, approaches, challenges and opportunities. Addis Ababa, Ethiopia: IRC.
- Milgrom, P. R., & Roberts, J. (1992). *Economics, organization, and management*. Englewood Cliffs, NJ: Prentice-Hall.
- Miller, G., & Hammond, T. (1994). Why Politics is More Fundamental Than Economics. Journal of Theoretical Politics, 6(1), 5-26.
- Montgomery, M. A., Bartram, J., & Elimelech, M. (2009). Increasing Functional Sustainability of Water and Sanitation Supplies in Rural Sub-Saharan Africa. *Environmental Engineering Science*, 26(5), 1017-1023. doi:10.1089/ees.2008.0388
- Montgomery, M. A., Bartram, J., & Elimelech, M. (2009). Increasing Functional Sustainability of Water and Sanitation Supplies in Rural Sub-Saharan Africa. *Environmental Engineering Science*, 26(5), 1017-1023. doi:10.1089/ees.2008.0388
- Moriarty, P., Smits, S., Butterworth, J., & Franceys, R. (2013). Trends in Rural Water Supply: Towards a Service Delivery Approach. *Water Alternatives*, 6(3), 329-349.
- Muggah, R. (2011). Urban Resilience in Situations of Chronic Violence; Case Study of Nairobi, Kenya. Institute of International Relations, Pontificia Universidade Católica do Rio de Janeiroand Center for Conflict Development and Peace, Geneva.
- Müller, J. (2016). *Harnessing private finance to attain public policy goals?* Amsterdam, Nertherlands: Centre for Research on Multinational Corporations (SOMO).
- Mwangi, T. W. (2013). Exploring Kenya's Inequality: Pulling Apart or Pooling Together? Kenya National Bureau of Statistics (KNBS) and Society for International Development (SID).

- Naughton, M., Deshmukh, R., Ahrari, S., & Gosling, L. (2018). Social accountability for rural water services. Rural Water Supply Network.
- Njonjo, K. S. (2013). Exploring Kenya's Inequality; Pulling Apart or Pooling Together?. Kenya National Bureau of Statistics (KNBS) and the Society for International Development (SID).
- OECD. (2011). OECD Studies on Water: A Multi-level Approach. Paris: OECD Publishing.
- Office of Evaluation and Oversight. (2016). Study on the Performance and Sustainability of Water and Sanitation Initiatives in Rural Areas. Inter-American Development Bank.
- O'Flynn, J. (2005, September). Adding public value: A new era of contractual governance? Paper presented at PAC Annual Conference-Public Administration and Management, University of Nottingham, UK.
- O'Flynn, J. (2014, November). Adding Public Value: A New Era of Contractual Governance? Paper presented at IPAA National Conference 2004, Canberra, Australia.
- Origa, J. (2017). Water schemes functionality crisis in Kenya: Policy recommendations for private partnership. Nairobi, Kenya.
- Ostrom, E. (2000). Social Capital: A Fad or a Fundamental Concept. In *Social Capital: A Multifaceted Perspective* (pp. 172-2014). Washington DC: World Bank.
- Oxfam International. (2016). Monitoring of strategic water points for drought early warning in the Horn of Africa. Nairobi, Kenya: Oxfam.
- Oyo, A. (2002). Creating Successful Private Sector Supply Chains. A resource guide for rural water supply and sanitation practitioners (46345). Retrieved from World Bank website: http://documents.worldbank.org/curated/en/588731468142783903/Creating-

successful-private-sector-supply-chains-a-resource-guide-for-rural-watersupply-and-sanitation-practitioners

PEMconsult, CESPA, & Colan Consult. (2013). Study on Management Options & Financial Sustainability for Small Town Water Supplies in Sierra Leone. Ministry of Water Resources Siera Leon.
- Perry, C. J., Rock, M., & Seckler, D. (1997). Water as an economic good: A solution, or a problem? Colombo, Sri Lanka: International Water Management Institute.
- Peter, G., & Nkambule, S. E. (2012). Factors affecting sustainability of rural water schemes in Swaziland. *Physics and Chemistry of the Earth, Parts A/B/C*, 50-52, 196-204. doi:10.1016/j.pce.2012.09.011
- Peters, B. G. (2001). The future of governing. Lawrence: University Press of Kansas.
- Pierre, J. (2006). *Debating governance: [authority, steering and democracy]*. Oxford, UK: Oxford University Press.
- Pierre, J., & Peters, B. G. (2010). Governance, Politics and the State. Palgrave Macmillan.
- Pritchett, L., & Woolcock, M. (2002). Solutions When the Solution is the Problem: Arraying the Disarray in Development. SSRN Electronic Journal.
- Republic of Kenya. (1999). Sessional paper no. 1 of 1999 on national policy on water resources management policy and development. Nairobi, Kenya: The Government Printer.
- Republic of Kenya. (2002). Laws of Kenya: The Water Act, 2002. Nairobi, Kenya: Government Printer.
- Republic of Kenya. (2010). The Constitution of Kenya. Nairobi, Kenya: Government Printer.
- Republic of Kenya. (2016). Laws of Kenya: The Water Act, 2016. Nairobi, Kenya: Government printer.
- Rockinson-Szapkiw, A. J., & Spaulding, L. S. (2014). Navigating the doctoral journey: A handbook of strategies for success. Lanham, MD: Rowman & Littlefield Publishers.
- Rogers, P, P., Bhatia, R., & Huber, A. (1999). Water as a social and economic good: How to put the principle into practice. Stockholm: SIDA - Swedish International Development Authority.
- Rogers, P. (2002). Water is an economic good: How to use prices to promote equity, efficiency, and sustainability. *Water Policy*, 4(1), 1-17.

- Rogers, P., & Hall, A. W. (2003). Effective water governance: Learning from the dialogues (TEC Background Papers Series No.7). Stockholm: Global Water Partnership.
- Rural Water Supply Network. (2010). *Myths of the rural water supply sector*. St. Gallen, Switzerland: Author.
- RWSN Executive Steering Committee. (2010). *Myths of the rural water supply sector* (RWSN Perspective No 4). Rural Water and Sanitation Network.
- RWSN Executive Steering Committee. (2010). Myths of the rural water supply sector. Rural Water Supply Network.
- Saurí, D., & Del Moral, L. (2001). Recent developments in Spanish water policy. Alternatives and conflicts at the end of the hydraulic age. *Geoforum*, 32(3), 351-362.
- Savenije, H. H., & Van der Zaag, P. (2002). Water as an Economic Good and Demand ManagementParadigms with Pitfalls. *Water International*, 27(1), 98-104.
- Schouten, T., & Moriarty, P. (2013). *The triple-S theory of Change*. Hague, Nertherlands: IRC International Water and Sanitation Centre.
- Schweitzer, R. W. (2009). Community managed rural water supply systems in the Dominican republic (Unpublished doctoral dissertation). Michigan Technological University, Michigan.
- Shah, A., & Mills, R. (2018). Funding Mechanisms to Incentivize Sustainable and Inclusive Water Provision in Kenya's Arid Lands. Oxford, UK: Oxfam GB.
- Sheng, Y. K. (n.d.). *What is Good Governance?* United Nations Economic and Social Commission for Asia and the Pacific.
- Shiva, V. (2001). *Water wars: Privatization, pollution, and profit*. Toronto, Canada: Between the Lines.
- Simukonda, K., Farmani, R., & Butler, D. (2018). Intermittent water supply systems: causal factors, problems and solution options. Urban Water Journal, 15(5), 488-500. doi:10.1080/1573062x.2018.1483522
- Skinner, J. (2009). Where every drop counts: Tackling rural Africa's water crisis. Retrieved from International Institute for Environment and Development website: http://www.jstor.org/stable/resrep01430

SNV. (2010). Study of rural water supply service delivery models in Vietnam. Author.

- SNV-Netherlands Development Organisation. (2010). *Study of rural water supply service delivery models in Vietnam*. The Hague, Nertherlands: Author.
- Stiglitz, J. E., & Rosengard, J. K. (2015). Economics of the public sector.
- Stoker, G. (1998). Governance as theory: five propositions. *International Social Science Journal*, *50*(155), 17-28.
- Sutton, S. (2004). Preliminary desk study of potential for self-supply in sub-Saharan Africa. SWL Consultants.
- Tabi, M. T., & Verdon, D. (2014). New public service performance management tools and public water governance: the main lessons drawn from action research conducted in an urban environment. *International Review of Administrative Sciences*, 80(1), 213-235. doi:10.1177/0020852313511574
- Travis, K., & Sara, J. (1997). Making rural water supply sustainable: Recommendations from a global study. Retrieved from UNDP-World Bank Water and Sanitation Program website: http://documents.worldbank.org/curated/en/358261468200668527/Making-

rural-water-supply-sustainable-recommendations-from-a-global-study

- Turkana County Government. (2016). *The Turkana County Water and Sewerage Services* Sector Policy. Author.
- Turkana county Government. (2016). The Turkana County Water and Sewerage Services Sector Policy, August 2016. Lodwar, Kenya: Author.
- UN System Task Team on the post-2015 development agenda. (2012). *Governance and development*. United Nations Development Programme.
- United Nations Development Programme. (1997). Governance for sustainable human development: UNDP policy document. New York, NY: Author.
- United Nations. (1992). *The Dublin Statement on Water and Sustainable Development*. Dublin, Ireland: Author.
- United Nations. (2010, July 28). UN News General Assembly declares access to clean water and sanitation is a human right. Retrieved December 15, 2017, from http://www.un.org/apps/news/story.asp?NewsID=35456#.WjLyinlx3IU

- United Nations. (2011). Towards human resilience: Sustaining MDG progress in an age of economic uncertainty. Place of publication not identified: United Nations Publications.
- United Republic of Tanzaia. (2016). Water Point status in Rural areas by Region. Ministry of Water and Irrigation.
- Van der Wal, Z. (2017). The 21st century public manager: Challenges, people and strategies. London, UK: Palgrave.
- Van Koppen, B. (2003). Water reform in Sub-Saharan Africa: what is the difference? *Physics and Chemistry of the Earth, Parts A/B/C*, 28(20-27), 1047-1053...
- Varady, R. G., Zuniga-Teran, A. A., Garfin, G. M., Martín, F., & Vicuña, S. (2016). Adaptive management and water security in a global context: definitions, concepts, and examples. *Current Opinion in Environmental Sustainability*, 21, 70-77.
- Walters, J. P., & Javernick-Will, A. N. (2015). Long-Term Functionality of Rural Water Services in Developing Countries: A System Dynamics Approach to Understanding the Dynamic Interaction of Factors. *Environmental Science & Technology*, 49(8), 5035-5043.
- Wanyande, P. (2016). Devolution and territorial development inequalities: The Kenyan experience. Santiago, Chile: Working Paper Series N° 187, Rimisp.
- Water Services Regulatory Board. (2018). A Performance Report of Kenya's Water Services Sector 2015 / 16 and 2016 / 17. Nairobi, Kenya: Author.
- Water Services Regulatory Board. (2018). Impact 10: A Performance Report of Kenya's Water Services Sector 2015 / 16 and 2016 / 17. Retrieved from Water Services Regulatory Board website: https://wasreb.go.ke/impact-report-issue-no-10/
- Weimer, D. L., & Vining, A. R. (2017). Policy analysis: Concepts and practice (6th ed.). New York: Routledge.
- Whaley, L., & Cleaver, F. (2017). Can 'functionality' save the community management model of rural water supply? Water Resources and Rural Development, 9, 56-66.
- Widmalm, S. (2015, March). After NPM, curb your enthusiasm for the Principal-Agent theory. Paper presented at Quality of Governance seminar, Gothenburg.

- World Bank Group. (2017). Sustainability Assessment of Rural Water Service Delivery Models : Findings of a Multi-Country Review. Washington DC, WA: World Bank.
- World Bank Group. (2018). Kenya Economic Update, September 2018. Author.
- World Bank. (2004). World Development Report 2004: Making Services Work for Poor People. S.I.: Author.
- World Bank. (2016, November). Global study on sustainable service delivery models for rural water. Paper presented at 7th Global Rural Water Supply Network Forum, Abidjan, Cote d'Ivoire.
- World Bank. (2017). Rural Water in Tanzania: High Investments, Low Returns. World Bank Group.
- World Economic Forum. (2015). *Global Risks 2015,10th edition* (REF: 090115). Retrieved from World Economic Forum website: http://www.weforum.org/risks
- World Health Organization(WHO), & United Nations Children's Fund (UNICEF). (2017). Progress on drinking water, sanitation and Hygiene: 2017 update and SDGs baselines. S.I.
- Young, R. A. (2012). *Determining the Economic Value of Water*. Washington, DC: Taylor & Francis.
- Zetland, D. (2014). *Living with water scarcity*. Amsterdam ; Mission Viejo ; Vancover: Aguanomics Press.

APPENDICES

APPENDIX A: ETHICAL CLEARANCE FOR DATA COLLECTION



4th December 2018

SU-IERC0278/18

James Origa OTIENO

P.O BOX 30577--0100 Nairobi Kenya

Email: james otieno@strathmore.edu

Dear James.

REF Student Number: MPPM/96532/2016 Protocol ID: SU-IERC0278/18 Effectiveness of revenue management and maintenance practices of water service providers for sustainable rural water services provision: A case of Turkana County, Kenya.

We acknowledge receipt of your application documents to the Strathmore University Institutional Ethics Review Committee (SU-IERC) which includes:

- 1. Cover letter dated October 31 2018
- Research Proposal dated October 2018 2.
- 3. Data collection tools dated October 2018
- 4. Participant information sheet and informed consent form

5. CV

The committee has reviewed your application, and your study "Effectiveness of revenue management and maintenance practices of water service providers for sustainable rural water services provision; A case of Turkana County, Kenya" has been granted approval.

This approval is valid for one year beginning 4th December 2018 until 3rd December 2019.

In case the study extends beyond one year, you are required to seek an extension of the Ethics approval prior to its expiry. You are required to submit any proposed changes to this proposal to SU-IERC for review and approval prior to implementation of any change.

SU-IERC should be notified when your study is complete.

Thank you Sincerely,

Balin Amina Salim **Regulatory Affairs Fellow**



Ole Sangale Rd, Madaraka Estate. PO Box 59857-00200, Nairobi, Kenya. Tel +254 (0)703 034000 Email info@strathmore.edu www.strathmore.edu

APPENDIX B: DATA COLLECTION TOOLS

Household questionnaires

SECTION I: LOCATION

Sub County.....

Ward/Location.....

Village.....

SECTION II: DEMOGRAPHICS

- 1. Gender
 - o Male
 - o Female
- 2. Age Bracket
 - o Less than 18years
 - o 18-30years
 - o 30-40years
 - o 40-50years
 - o More than 50years
- 3. What is the highest level of education attained?
 - Never went to school
 - o Didn't complete primary School
 - o Primary School
 - o Didn't complete secondary School
 - Secondary School
 - Technical School
 - o University

AFFORDABILITY AND WILLINGNESS TO PAY

	4.	Do you pay for water fetched from the source for domestic and livestock use?				
		o Yes				
		o No				
5.		f yes, how do you pay				
		• Pay as you fetch				
		 Fixed Monthly contribution 				
		 Ad hoc payment; only when asked/needed. 				
		• Other (Specify				
	6.	much do you pay on average?				
		o 20 litre containers				
		 For a Fixed Monthly contribution 				
		 For an Ad hoc payment only when asked/needed 				
		• Other (Specify				
		b) How much do you pay for livestock?				
		o Camel				
		o Cattle				
		o Shoats				
		 Others (Specify) 				
	7.	Do you find the water tariff/fees affordable?				
		o Yes				
		o No				
	8.	If No, how much would you have wanted to pay for water?				
		c) 20 litre container				
		d) For a Fixed Monthly contribution				
		e) For an Ad hoc payment only when asked/needed				
		f) Camel				
		g) Cattle				
		h) Shoats				
		i) Others (Specify)				
	9.	Are you willing to pay in future?				
		o No				
		• Yes- only per 20 litres				
		• Yes- only fixed monthly contribution				
		• Yes-only when asked in case of breakdown				
		• Yes- per bucket and in case of breakdown				
	10	o Uther (Specify)				
	10.	What happens if you cannot afford the price of water?				

o Fetch from alternative unsafe water sources

.

- o Fetch on credit
- Pay using livestock
- o Borrow from neighbours/friends
- o Other (Explain).....

MAINTENANCE AND FUCNTIONALITY

- 11. a) Is the water point currently functional?
 - o Yes
 - o No
 - b) If no, why is it not functional?
 - o It has dried up
 - o There's a breakdown
 - • There's a conflict over the water point
 - Other (Specify)

12. What is the number of days the water point has been functional in the last 2 weeks (14 days)

- o 12-14 days
- o 8-11 days -
- o 4-7 days
- o Less than 4 days

13. Has your water point experienced any breakdown in the last two weeks?

- o Yes
- o No

14. a) If yes, what caused the breakdown?

- Broken pipes and taps
- Fault in the pump
- o Faulty storage tank
- o No fuel
- o Lack of service
- o Any other reason

15. How long does it normally take to repair the facility in case of any breakdown?

- o Less than 3 days
- o 3 days
- o Less than a day
- o More than a week
- More than a month
- o Don't know

16. During breakdown of the facility, are households asked to contribute towards the repair?

- o Yes
- o No

17. Who is involved in carrying out the repairs?

- County government engineers/technicians
- o NGO's
- o Water Management committee members
- o Technician hired from town centres (Lodwar, Kakuma, etc.)
- o County water companies e.g. LOWASCO, WAJWASCO
- o Others, specify

18. In your opinion who should pay if the water facility breaks down?

- o Government because water is a gift
- o Water Management Committee because it is their responsibility
- Water Service Provider/Company
- o Myself; because water is something we buy
- o NGOs
- o Any other, specify
- **19.** In your opinion, are the funds collected from water use sufficient to cover the costs of operations and maintenance?
 - o Yes
 - o No
- 20. If No, what additional sources can support O & M?

FOCUSED GROUP DISCUSSIONS GUIDE FOR THE WATER MANAGEMENT COMMITTEES

Name of Community/water point:

.....

Date of Interview:

***************************************	•
Name of	

Interviewer.....

•••••

1. What are the key responsibilities of the committee on issues relating to management of the water facilities?

.....

Collection and management of revenue needed for the daily operations of the water facility

2. Do you collect any water use tariff from the water users?

(a) Yes (b) No

3. If no, how do you raise revenue needed for O & M?

4. If yes, how was the tariff decided?

(a) By the committee

(b) By the community and the committee

(c) By the committee and community representatives

(d)others (specify).....

5. Are people willing and able to pay the rate?

(a) Yes (b) No

If no, why?

.....

6. What are the collected revenues used for?

Expenditure item	Amount used (Ksh.)
Buy fuel	
Buy spare parts (specify which ones)	
Pay operations staff (list different staff and how much they are paid)	
Annual maintenance	
Response to breakdown	
List all other expenses	

7. Is the revenue information/report(s) shared with the public for review and discussions?(a) Yes (b) No

c) If yes, how is this information dispatched to the public?

......

.....

8. Does the community make contributions to how the revenue should be spent?

(a) Yes (b) No

9. Are the revenues raised sufficient to cover all the costs/expenditures of O & M?

(a) Yes (b) No

10. If no, how do you fill in the gap of revenues needed for O & M?

.....

.....

Maintenance

11. How often do you carry out the various maintenance activities on the water facilities?

List Maintenance activity/type	daily	weekły	monthly	Quarterly	yearly	Other(specify)

12. How long does it take to respond to breakdown/failure?

.....

13. How do you access/ get spare parts supplies to maintain the water facility? Are there any challenges in accessing the spare parts?

.....

.....

14. What is the relationship between you and the following institutions on the management of water facilities:

Institution	Form/Nature of collaboration/Association				
	Technical support	Financial support	Others (specify)		
County water department	2 				
NGOs (name them)					
Contractors					

Local mechanics/plumbers		
Traditional elders/authority		
Politicians/MCA/MP etc.		
Others (specify)		

15. In general, what problems/challenges do you encounter in ensuring management of water facilities?

.....

16. What measures do you suggest could be adopted to address these problems/challenges?

INTERVIEW GUIDE OF THE COUNTY GOVERNMENT OFFICIALS AND NGOS

Designation of Respondent:
Date of Interview:
Name of Interviewer

1. How is the County government/NGO involved in the daily operations and maintenance and mobilization of revenues needed for the maintenance of rural water facilities in the county?

.....

- 2. Is the revenue collected by water management committees sufficient to cover the full costs of operations and maintenance.
- 3. If no to question 3, what role does the county government/NGO has in ensuring sufficiency of funds to cover operations and maintenance needs.....
- 4. Does the county government/NGO offer technical support and monitor the operations of the service providers managing the rural water points within the county? How is the support delivered/structured?
- 5. What other problems do you encounter in the operations and maintenance of rural water facilities in the county?

6. What measures do you suggest could be used to address the problems mentioned above?

- -----
- 7. Any further suggestions/comments

....?

.....